brought to you by T CORE

MEASURING BANK COMPETITION IN CHINA: A COMPARISON OF NEW VERSUS CONVENTIONAL APPROACHES APPLIED TO LOAN MARKETS

Bing Xu, Adrian van Rixtel and Michiel van Leuvensteijn

Documentos de Trabajo N.º 1404

BANCODEESPAÑA

Eurosistema



MEASURING BANK COMPETITION IN CHINA: A COMPARISON OF NEW VERSUS CONVENTIONAL APPROACHES APPLIED TO LOAN MARKETS

MEASURING BANK COMPETITION IN CHINA: A COMPARISON OF NEW VERSUS CONVENTIONAL APPROACHES APPLIED TO LOAN MARKETS ⁽¹⁾

Bing Xu

UNIVERSIDAD CARLOS III

Adrian van Rixtel (**)

BANK FOR INTERNATIONAL SETTLEMENTS

Michiel van Leuvensteijn

ALL PENSIONS GROUP

(*) Bing Xu, Economics Department, Universidad Carlos III de Madrid, Calle Madrid 126, 28903 Getafe, Spain, bxu@eco.uc3m.es or xuduan_nju@hotmail.co.uk. Adrian van Rixtel, Financial Markets Group, Monetary and Economic Department, Bank for International Settlements (BIS), Centralbahnplatz 2, 4002 Basel, Switzerland, tel. +41612808252, adrian.vanrixtel@bis.org. Michiel van Leuvensteijn, Corporate Strategy & Policy, APG, Gustav Mahlerplein 3, 1082 MS Amsterdam, The Netherlands, tel. +31206049173, michiel.van.leuvensteijn@apg.nl. We are grateful to Yiping Huang and Xun Wang for sharing the values of their financial repression index with us. We would like to thank an anonymous referee, Claudio Borio, Chun-Yu Ho, Tom Minic, Christian Upper and participants at a BIS seminar, at the "VIII Annual Seminar on Risk, Financial Stability and Banking", organised by Banco Central do Brasil (8-9 August 2013, São Paulo), and at the "Fifth Annual International Conference on the Chinese Economy", organised by the Hong Kong Institute for Monetary Research (16-17 January 2014, Hong Kong) for helpful comments. All remaining errors are ours. The views expressed in this paper are the authors' and do not necessarily reflect those of the BIS or APG. (**) Bank for International Settlements. Corresponding author.

Documentos de Trabajo. N.º 1404 2014

The Working Paper Series seeks to disseminate original research in economics and finance. All papers have been anonymously refereed. By publishing these papers, the Banco de España aims to contribute to economic analysis and, in particular, to knowledge of the Spanish economy and its international environment.

The opinions and analyses in the Working Paper Series are the responsibility of the authors and, therefore, do not necessarily coincide with those of the Banco de España or the Eurosystem.

The Banco de España disseminates its main reports and most of its publications via the INTERNET at the following website: http://www.bde.es.

Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged.

© BANCO DE ESPAÑA, Madrid, 2014

ISSN: 1579-8666 (on line)

Abstract

Since the 1980s, important and progressive reforms have profoundly reshaped the structure of the Chinese banking system. Many empirical studies suggest that financial reform promoted bank competition in most mature and emerging economies. However, some earlier studies that adopted conventional approaches to measure competition concluded that bank competition in China declined during the past decade, despite these reforms. In this paper, we show both empirically and theoretically that this apparent contradiction is the result of flawed measurement. Conventional indicators such as the Lerner index and Panzar-Rosse H-statistic fail to measure competition in Chinese loan markets properly due to the system of interest rate regulation. By contrast, the relatively new Profit Elasticity (PE) approach that was introduced in Boone (2008) as Relative Profit Differences (RPD) does not evidence these shortcomings. Using balance sheet information for a large sample of banks operating in China during 1996-2008, we show that competition actually increased in the past decade when the PE indicator is used. We provide additional empirical evidence that supports our results. We find that these, firstly, are in line with the process of financial reform, as measured by several indices, and secondly are robust for a large number of alternative specifications and estimation methods. All in all, our analysis suggests that bank lending markets in China have been more competitive than previously assumed.

JEL classification: D4, G21, L1.

Keywords: competition, banking industry, China, lending markets, marginal costs, regulation, deregulation.

Resumen

Desde los años ochenta, la estructura del sistema bancario chino se ha visto reconfigurada a través de importantes reformas progresivas. Muchos estudios empíricos sugieren que la reforma financiera ha fomentado la competencia bancaria en la mayor parte de las economías avanzadas y emergentes. No obstante, algunos estudios previos que adoptaron enfoques convencionales para medir la competencia concluyeron que la competitividad bancaria en China descendió durante la década pasada, a pesar de estas reformas. En el presente análisis se muestra de forma teórica y empírica que esta aparente contradicción se debe a una medición errónea. Diversos indicadores convencionales - como el índice Lerner y el estadístico-H de Panzar-Rosse - no logran medir apropiadamente la competencia en los mercados de préstamo chinos, debido al sistema de regulación de los tipos de interés. Por el contrario, el enfoque relativamente novedoso de la elasticidad del beneficio (Profit Elasticity, PE) introducido por Boone (2008) a través de diferencias relativas en los beneficios (Relative Profit Differences, RPD) no sufre estas deficiencias. Mediante la utilización de información procedente de los balances de una gran muestra de bancos operativos en China durante 1996-2008 se muestra que, en realidad, la competencia se ha visto incrementada durante la pasada década, cuando se emplea el indicador PE. Se proporciona una evidencia empírica adicional que apoya nuestros resultados. En primer lugar, se confirma que estos se encuentran alineados con el proceso de reforma financiera, medido según diversos índices; y, en segundo lugar, que son robustos en un gran número de especificaciones alternativas y métodos de estimación. En términos generales, nuestro análisis sugiere que los mercados de préstamos bancarios en China han sido más competitivos de lo que se asumía en la literatura.

Códigos JEL: D4, G21, L1.

Palabras clave: competencia, sector bancario, China, mercados de crédito, costes marginales, regulación, desregulación.

1 Introduction

In recent years, the Chinese banking sector has become the focus of a growing body of empirical literature. Studies have tried to measure developments in bank efficiency and bank productivity, the contribution of bank intermediation to China's economic growth and particular characteristics of Chinese banks' corporate governance structures. Notwithstanding this progress in empirical research, relatively few econometrical analyses have concentrated explicitly on bank competition in China.

The apparent lack of these studies is somewhat surprising, for a number of reasons. First, although different views persist, several established observers have identified competition as one of the key factors in China's economic reform success. They argue that the expansion of incentives, mobility and markets has created unprecedented business opportunities, exemplified in astonishingly high scales of entry and exit. This process has pushed China's economy towards "extraordinarily high levels of competition", leading to a situation where "intense competition now pervades everyday economic life" (Brandt and Rawski, 2008).¹ These claims make one wonder whether this general picture can be extended to a specific sector such as banking as well.

Second, with China becoming one of the major global economic powers, overtaking Japan as the world's second-biggest economy at the end of 2010, also Chinese banks have become global powerhouses. The "Big Four" state-owned commercial banks (SOCBs) are now among the ten largest banks in the world according to market capitalisation, and China's banks accounted for 20% of global banking profits in 2010 (KPMG, 2011; Löchel and Li, 2011). They have embarked upon a global expansion strategy, opening new branches and subsidiaries abroad and forming cross-border alliances in bank services, while also developing internationally more diversified business lines in insurance and asset management. These global business advances originate from domestic market conditions, including certain degrees of efficiency, contestability and competition, providing additional arguments for investigating the latter.

Third, the process of financial reform has reshaped China's banking sector profoundly. The prevailing view is that after 30 years of financial reform, China's banks have begun to behave more like commercial banks in the developed world (Firth et al., 2009). During this period, the banking landscape in China has changed dramatically. In essence, it moved from a socialist monolithic structure to a pluralistic system comprising various groups of market-oriented banks. The latter development started with the creation of a "two-tier" banking sector in 1978 with the establishment of the People's Bank of China (PBC) as the central bank and four large state-owned specialised banks serving specific sectors of the economy. It took 16 years before the Chinese government started a new major round of reforms, when in 1994 three specialised policy banks were established to take over the policy loans from the "Big Four" banks and the status of the latter was changed to state-owned commercial banks. The conditions for China's WTO accession and its actual entry into this organisation in 2001 triggered a major third round of reforms, including the establishment of an independent bank supervisor (China Banking Regulatory Commission or CBRC) and

^{1.} This despite China being a clear counter-example with respect to the conclusions from the existing literature on law, institutions, finance and growth on the advocated institutional framework conducive to economic development (Allen et al., 2005).

further measures regarding the liberalisation of interest rates, business scope and market entry. Overall, while acknowledging remaining deficiencies, the Chinese banking sector has moved gradually but unmistakably towards a commercially oriented market-type system, in which competitive forces should take on much greater significance.

Last but certainly not least, investigation of bank competition in China contributes to the continuing debate on which empirical approaches may be the most suitable for measuring competition in specific banking systems. This argument has assumed growing weight in the bank competition literature, underpinning the rather unsatisfactory observation that the currently available empirical toolkit frequently yields contradictory and inconclusive results for specific countries and/or regions. For example, one study concludes that "well-known indicators of bank competition often give conflicting predictions of competitive behaviour across countries, within countries, and over time" and that the "determination of competition may differ depending on the measure chosen to assess it" (Carbó Valverde et al., 2009, p. 132). These findings suggest that it may be preferable to consider different measures when assessing bank competition.

This paper contributes to the literature on both Chinese banking and bank competition by arguing that conventional measures of competition like the Lerner index and the Panzar-Rosse H-statistic may not assess bank competition in China correctly, mainly due to the existence of interest rate regulations. The former uses the profit or price cost margin (PCM), ie the mark-up in output prices above marginal cost, as an indicator of market power. The latter measures to what extent input and output prices move in step (as they would under perfect competition) or out of step (indicating monopoly or a perfect cartel) (Bikker, 2010). The shortcomings of these approaches in the context of Chinese banking markets can be summarised as follows. First, the Lerner index and Panzar-Rosse H-statistic may bias the results due to the system of interest rate regulation in China. Second, several other characteristics of Chinese banking make these conventional measures less appropriate as well, in our view.

Instead, we argue that the relatively new Profit Elasticity (PE) indicator may be better suited to investigate competitive conditions in Chinese loan markets, given the particularities of the banking industry in China. This indicator, whose theoretical base is the Relative Profit Differences (RPD) concept, is based on the idea that competition rewards efficiency (Boone et al., 2007; Boone, 2008; Van Leuvensteijn et al., 2011 and 2013). In general, an efficient firm will realise higher profits than a less efficient one. Crucial for the PE indicator is that this effect will be stronger the more competitive the market is.² This can be explained as follows. In the theoretical setup of RPD, competition increases due to increased interaction between banks or due to lower entry costs. Boone (2008) shows that RPD is an increasing function of the degree of interaction between firms and a decreasing function of entry costs. Hence, RPD increases when competition intensifies, ie stronger competition decreases (increases) profits of more efficient firms by smaller (larger) amounts than for less efficient firms. The underlying intuition is that in a more competitive market, firms are punished more harshly (in terms of profits) for being inefficient. All in all, the PE indicator is a new measure of competition that is more robust from both a theoretical and an empirical point of view when compared with more conventional measures (Boone et al., 2007). Early empirical applications of the PE indicator are Van Leuvensteijn et al. (2007), Van Leuvensteijn (2008) and Bikker and Van Leuvensteijn (2008), while a more explicit empirical validation has been provided by Boone and Van Leuvensteijn (2010).

^{2.} As with all measures of competition, the PE indicator is also based on certain assumptions, such as that of product homogeneity (product innovation does not matter). These assumptions will be discussed in more detail in Section 4.2.

The paper contributes to the existing literature in a number of ways. It combines, according to our knowledge, for the first time the Lerner, Panzar-Rosse and PE approaches in one paper.³ We also demonstrate theoretically that RPD, from which the PE indicator is derived, is much better suited to assess competition in banking markets where interest rates are regulated, whereas in that case the Lerner index yields biased results. Furthermore, we assess the relationship of the various approaches with the process of financial reform. All in all, the paper provides an extensive discussion of the appropriateness of competition indicators for a specific country, in this case China, and argues that banking system specifics may significantly affect the results of different indicators of competition. Hence, we warn against "auto-pilot" applications of empirical measures to assess banking competition across countries. Moreover, the paper yields important new results regarding the development of competition in Chinese loan markets.

These results can be summarised as follows. First, the conventional measures - both the Lerner index and the Panzar-Rosse H-statistic - show declining competition over time in Chinese loan markets. This despite the comprehensive process of financial reform, which according to a large body of empirical research promoted banking competition in many mature and other emerging market economies. The results for both indicators hold for alternative specifications, suggesting that they in themselves are estimated correctly. Second, we find that the long-run equilibrium assumption on which the H-statistic depends is violated, also for alternative specifications. This may be related to the financial reform process which continues to reshape the banking landscape in China profoundly. Hence, inferring competitive conditions from it for China is likely to give rise to bias (apart from in our view fundamental biases inherent in the Panzar-Rosse model with respect to China). Third, in contrast, the findings for the PE indicator show improving competition in Chinese loan markets over time, especially after 2001, with some retreat in the final years of our sample. Moreover, we are fairly able to explain the specific pattern of the development of competition. Fourth, our results for the PE indicator are in line with the development of various indicators of financial reform. Finally, the findings for the PE indicator are robust for several alternative specifications and pass various robustness tests. All in all, we see our a priori theoretical objections to the Lerner index and Panzar-Rosse H-statistic as appropriate measures to assess Chinese banking competition validated by the empirical results. The results support our belief that the PE indicator is better suited for this purpose. Overall, we contribute to both the ongoing discussion in the banking competition literature on the appropriateness of different competition measures and the growing empirical body of research on Chinese banking.

The remainder of this paper is organised as follows. Section 2 provides a brief overview of the research literature on empirical measures of bank competition. Section 3 first gives background information on the structure of Chinese banking (3.1), followed by a review of China-specific studies on bank efficiency and competition (3.2). Section 4 presents the methodological framework of the standard and elasticity-adjusted Lerner indices (4.1) and the PE indicator (4.2). Moreover, it demonstrates that the Lerner index yields biased results under binding interest rate regulation and RPD not (4.3). Section 5 shows our data and sample characteristics. Section 6 presents the empirical results for the (elasticity-adjusted) Lerner index (6.1) and the PE indicator (6.2). It then compares the results of the various empirical measures (6.3) and presents a detailed interpretation of the results from the PE indicator, including their relationship with various financial reform indicators (6.4). Section 7 concludes.

^{3.} The Panzar-Rosse H-statistic is presented in Appendix A.

2 Review of empirical literature on bank competition

Competition in the banking sector has generally been analysed on the basis of two concepts, ie market power and efficiency, which have sometimes been tested jointly. Here, market power reflects the ability of specific banks to control the market for bank products, whereas efficiency relates to the ability of specific banks to produce output (such as loans) at minimal cost.

With respect to the first concept for investigating bank competition, a well-known approach to measuring market power is suggested by Bresnahan (1982) and Lau (1982). These authors analyse bank behaviour on an aggregate level and estimate the market power of an average bank following a specific short-run model. Empirical studies based on this approach are rather scarce though, as it is very data-intensive. Another approach based on market power has been proposed by Panzar and Rosse (1987), which is the so-called "Hstatistic" and which will be discussed in greater detail in Appendix A. The specific value of the "H-statistic" measure indicates how competitive the market is, ranging from a monopoly (or perfect collusion) to a situation of perfect competition. A third indicator for market power is the "Hirschman-Herfindahl Index" (HHI), which measures the degree of market concentration. This indicator is often used in the context of the "Structure Conduct Performance" (SCP) model, which assumes that market structure affects banks' behaviour, which in turn determines their performance. The idea is that a highly concentrated banking sector (with a few banks occupying significant market shares) can impair competition, in the sense that concentration translates into greater market power, resulting in collusive behaviour and excess profits for banks. Finally, market power may also be related to profits, in the sense that extremely high profits may be indicative of a lack of competition. A traditional measure of profitability is the price-cost margin (PCM), which is equal to the output price minus the marginal costs. The PCM is frequently used in the empirical industrial organisation literature as an empirical approximation of the Lerner index (see Section 4.1) and in this context applied to the banking sector as well.

With respect to the second concept for measuring bank competition, ie efficiency, this indicator is often seen as a proxy for competition, in the sense that the most efficient banks (and therefore the most competitive ones) will gain market share at the cost of the less efficient banks. A relatively recent method, which has been named the profit elasticity (PE) model or the "Boone" indicator, can be seen as an elaboration on this efficiency hypothesis. This measure has gained considerable support more recently (Van Leuvensteijn et al., 2007, 2011 and 2013; Van Leuvensteijn, 2008; Schaeck and Cihák, 2010; Delis, 2012; Tabak et al., 2012). The underlying model will be explained in more detail in Section 4.2.

The actual literature on the measurement of competition is generally categorised into two major streams (Bikker, 2004; Tabak et al., 2012). So-called structural approaches are based on the SCP model and use concentration indicators as proxies for competition, such as the HHI and the CRn which measures the market shares of the n largest banks. In contrast, non-structural approaches have been promoted within the so-called New Empirical Industrial Organisation (NEIO) literature. They estimate parameters that reflect the degree of competition in specific markets based on bank-level data and specific assumptions on the behaviour of banks. The Bresnahan, Lau and Panzar-Rosse approaches mentioned above, as well as the Lerner index and PE elasticity, fall into this part of the literature. While these measures have been broadly accepted, there is no consensus regarding which is the "best" indicator for gauging bank competition (Carbó Valverde et al., 2009). As a matter of fact, they often reach different conclusions for banking systems of the same countries and groups of countries. Consequently, leading experts in the field acknowledge that the ability of empirical research to capture the degree of bank competition is still imperfect and that developing proper competitiveness tests and methodologies will remain an important area of research (Claessens, 2009, p. 95; Claessens and Laeven, 2004, p. 581).

At the same time, a view that has been gaining support is that concentration may not be the most appropriate indicator for measuring bank competition (Bikker, 2004; Casu and Girardone, 2006 and 2009; Schaeck et al., 2009). Moreover, concentration does not measure the competitive conduct of banks at the margin. In addition, concentration indices such as the HHI do not distinguish between small and large countries and may incorrectly suggest that competition is declining, while in fact concentration and competition are increasing simultaneously as a result of bank consolidation (Van Leuvensteijn, 2008; Van Leuvensteijn et al., 2013).

One of the main issues that has been at the core of empirical studies on bank competition, and which may be especially relevant for China, is the impact of financial reform on the degree of competitiveness of banking systems and subsequent efficiency and welfare gains. There is considerable empirical evidence that deregulation fostered competitive conditions in banking markets and led to greater product differentiation, lower cost of and improved access to financial services and greater financial stability (Claessens, 2009). These effects have been validated empirically especially for the United States and emerging market economies, while results have been less conclusive for Europe.

3.1 The structure of the Chinese banking industry⁴

China's financial system has undergone a comprehensive process of reform during the past 30 years, of which one of the main objectives was to improve competition and efficiency in the banking sector.⁵ In this section, we provide an overview of developments which are of particular relevance for the investigation of competition in Chinese Ioan markets. These are the start of commercial banking in China, the entrance of new players, which was promoted by China joining the WTO in 2001, and the deregulation of the credit control system and of interest rates.

The Commercial Bank Law of the People's Republic of China was promulgated in May 1995, which paved the way for the development of a commercial banking system in China and the entrance of important new players (Fu and Heffernan, 2009). In this context, 12 so-called joint-stock commercial banks (JSCBs) and more than 100 city commercial banks (CCBs) were established. The former initially offered banking services only regionally, but later they were allowed to operate freely nationwide, competing with the state-owned commercial banks (SOCBs) for the large firms and with the CCBs for small and medium-sized enterprises. CCBs offer commercial banking services to small and medium-sized enterprises and households in the main cities or in certain provinces, but have been expanding to larger companies that would normally do business with the SOCBs and JSCBs. The requirement for CCBs to operate only within the cities' own administrative districts was lifted from 2007 onwards, allowing them to compete in larger geographical areas (Sun and Yamori, 2011).

The commercialisation of Chinese banking was triggered by mounting problems at the four state-owned specialised banks which concentrated on providing loans to stateowned enterprises in specific sectors. These "Big Four" (Bank of China, Agricultural Bank of China, China Construction Bank, and Industrial and Commercial Bank of China) experienced a significant deterioration of their asset quality in the early 1990s. This was reflected in a rapid increase of their non-performing loans, which consisted predominantly of loans to stateowned enterprises, granted mainly for political instead of business reasons. Three specialised policy banks were established to take over the policy-lending business of the "Big Four" banks. The latter were converted into state-owned commercial banks (SOCBs), and four asset management companies were given the task of absorbing their non-performing loans. Hence, since the mid-1990s, three main groups of Chinese commercial banks - SOCBs, JSCBs and CCBs - have become active in Chinese loan markets. Arguably, this expansion of the number of providers of credit may have promoted competitive conditions in these markets. In fact, as is shown in Table 1, the market share of the SOCBs has declined significantly. While their average annual market shares of total assets and loans during 1996-2001 were 86% and 88%, respectively, these shares dropped to 72% and 71%, respectively,

^{4.} This section will only pay attention to the major banks which are covered in our study and ignore smaller banks such as rural credit co-operatives. In the same vein, only those aspects of financial reform that are related to the banks in our sample will be discussed.

^{5.} According to the China Banking Regulatory Commission (CBRC), financial reform in China can be classified in three major stages (1978-1993, 1994-2002, 2003-present), which are discussed extensively in Liu (2009). Clear overviews in English are presented in, for example, Matthews and Zhang (2010) and Chang et al. (2012).

during 2002-2008.⁶ These declines in market shares have been mirrored in considerable increases in those of especially the JSCBs and of the CCBs as well.

Table 1

	9	Share of to	otal asset	ts (%)	1	Share of t	otal loan	s (%)	Share of total deposits (%)			
	SOCB	JSCB	CCB	FOREIGN	SOCB	JSCB	CCB	FOREIGN	SOCB	JSCB	ССВ	FOREIGN
Average 1996–2001	86.32	11.83	1.69	0.16	88.03	10.54	1.29	0.14	87.55	11.41	0.90	0.14
Average 2002–2008	72.17	21.25	5.78	0.79	71.30	22.40	5.43	0.87	74.47	21.51	3.43	0.59
Average 1996–2008	78.70	16.91	3.89	0.50	79.02	16.93	3.52	0.53	80.51	16.85	2.26	0.38
	Sh	are of tot	al securit	ies (%)	Share of pre-tax profits (%)				Return on assets (%)			
	SOCB	JSCB	ССВ	FOREIGN	SOCB	JSCB	ССВ	FOREIGN	SOCB	JSCB	ССВ	FOREIGN
Average 1996–2001	81.16	15.62	3.15	0.06	71.21	23.83	4.52	0.43	0.16	0.70	1.03	0.99
Average 2002–2008	76.45	17.20	5.95	0.41	71.85	21.19	5.90	1.05	0.72	0.45	0.63	0.93
Average 1996–2008	78.62	16.47	4.66	0.25	71.77	21.54	5.72	0.97	0.46	0.57	0.82	0.96

Overview of the Chinese banking sector 1996–2008

Source: BankScope, authors' own calculations. The data are presented for the full sample of 1996–2008 and the two subsamples that we use in the estimations (1996–2001 and 2002–2008), with 2001 being the year of China's entry in the WTO.

Competition may also have benefited from the growing role of foreign banks. An important catalyst here was China's accession to the WTO in 2001. Under the conditions of WTO membership, the activities of foreign banks were liberalised profoundly. For example, these banks were allowed to provide foreign currency services to Chinese residents and were permitted greater freedom in local currency operations as well. Furthermore, the participation of foreign investors in Chinese banks was promoted, with foreigners being allowed take equity stakes of up to 25%. Chinese banking markets have been opened to foreign competition rather comprehensively since the end of 2006, with foreign banks receiving in principle the same regulatory treatment as their Chinese counterparts and expanding their business further into the rest of the country (Yao et al., 2008; Matthews and Zhang, 2010; Xu, 2011). The foreign liberalisation of Chinese banking resulted in a sharp increase in the number of foreign players, whose number increased in our sample from four in 1996 to 26 in 2008 (see Section 5). Table 1 shows that the market shares of foreign banks in total assets and loans increased as well during 1996-2008, but remained below 1%. Their share in pre-tax profits rose to an annual average of above 1% in 2001-2008, reflecting a return on assets that has been the highest of all banking groups.

Despite the relatively small market shares of foreign banks in China, in our view the importance of their role in Chinese banking should not be underestimated. Namely, it has been argued rather widely that the impact of foreign banks on bank efficiency and competition may have significantly exceeded what their modest presence in China may suggest. First, various studies suggest that both the threat of foreign entry and its actual realisation forced Chinese banks to respond in terms of improving their business models, efficiency, market practices and hence their degree of competitiveness (He and Fan, 2004; Leung and Chan, 2006). Second, foreign banks have obtained minority equity stakes in various Chinese banks, which overall seem to have had positive effects on the efficiency and/or performance of the latter and hence likely on competition in Chinese banking markets

^{6.} We present the data for the full sample of 1996-2008 and the two subsamples that we use (1996-2001 and 2002-2008), with 2001 being the year of China's entry into the WTO.

(Berger et al., 2010). Third, foreign banks have tried to penetrate into other parts of China beyond the main cities through equity partnerships or less institutionalised forms of cooperation with Chinese banks with either geographical significance or appreciable levels of national coverage (He and Fan, 2004; Leung and Chan, 2006). Fourth, some observers claim that foreign banks have demonstrated that they are capable of snatching significant market shares away from Chinese banks, such as in RMB loan and deposit markets in key cities such as Beijing, Shanghai and Guangzhou (Xu and Lin, 2007). Finally, Xu (2011) provides empirical evidence that foreign bank entry into China has been supportive of developing a more competitive banking industry. All in all, we believe that these arguments support the inclusion of foreign banks in empirical investigations of bank competition in China.

Another important reform affecting Chinese loan markets was the replacement of the PBC's binding credit plan system with an indicative non-binding credit target, effective from 1 January 1998, with this target serving only as a reference for commercial banks (Mo, 1999). Until then, the PBC had controlled the lending of SOCBs through binding credit quotas, which set the lower limit for new loans to be made annually and stipulated their allocation to specific sectors (Wong and Wong, 2001). Hence, since 1998, in principle Chinese banks have become free to lend according to commercial considerations, with the formal abolishment of policy loans that were provided in compliance with state directives or planning targets instead of on the basis of proper credit assessments. This change in policy has been hailed by Chinese monetary authorities as an important step in transforming the credit culture of Chinese banks.

Notwithstanding the significance of the abolishment of the credit plan system, there are clear signs of continuing quantitative controls on bank credit, which potentially may affect the lending policies of banks in China. Various observers emphasise the use by the PBC of quantitative instruments aimed at controlling credit growth, despite the discontinuation of the binding credit plan system (Goodfriend and Prasad, 2006; Liu and Xie, 2006; Delatte, 2007; Geiger, 2008; Du, 2010; Fukumoto et al., 2010; Huang et al., 2010; He and Wang, 2011 and 2012; Ma et al., 2011; Chen et al., 2012; Martin, 2012; World Bank, 2012). These include yearly aggregate target levels for new loans and the use of so-called window guidance to influence the development of bank lending. The latter policy can be described as a form of moral suasion aimed at controlling in principle the sectoral direction of lending, although it is suspected that in practice the guidance has also affected the amount of lending (Green, 2005; Okazaki, 2007).

The reform of the credit control system has been followed – in terms of degree of deregulation – by interest rate liberalisation, resulting in relatively liberalised bank interest rates in 2004, when the deposit rate floor and the lending rate ceiling were eliminated for the major banks (Figure 1, left-hand panel).⁷ The liberalisation of the ceiling on the lending rate and the floor on the deposit rate in October 2004 implied that Chinese banks benefited from a more or less guaranteed minimum interest rate spread (due to the remaining floor on the lending rate and ceiling on deposit rate), while they faced no restrictions with respect to its potential maximum width (García-Herrero et al., 2005). The lending rate floor should inhibit competition to some extent, although probably not in a binding manner.⁸ In fact, during December 2004 and

^{7.} The PBC started to widen the floating band on banks' interest rates from 1998 onwards, after it liberalised interbank interest rates. Hence, it gave commercial banks more discretion in setting loan rates (People's Bank of China, 2005; Feyzioğlu et al., 2009).

^{8.} Fu and Heffernan (2009) mention that Chinese banks have always had some discretion over certain loan rates. For example, from 1999, small business loans could carry a premium of up to 30% (50% for the rural credit cooperatives) over the central bank rate.

December 2008 only between 19% and 29% of all loans were made at the floor lending rate, suggesting that most loan rates were higher (Figure 1, right-hand panel). In contrast, empirical research has suggested that the ceiling on deposit rates has been binding and put them at a level below equilibrium (Feyzioğlu et al., 2009; He and Wang, 2011; Ma et al., 2011).



The liberalisation of both credit controls and interest rates has had important positive repercussions for Chinese banks' lending policies. Many observers adhere to the view that these policies have become more market-oriented, characterised by increasing attention to risk management and loan monitoring and by greater diversification into less traditional areas such as consumer lending (Allen et al., 2005; Leung and Chan, 2006; Abiad et al., 2010; Herd et al., 2010; Qian et al., 2011; IMF, 2011). In principle, this development should have been supportive of enhancing competitive conditions in Chinese loan markets.

This development has been aided by the establishment of the China Banking Regulatory Commission (CBRC) in 2003, which helped bring lending policy more into line with market-conform assessment and approval criteria (Yeung, 2009). Moreover, since October 2004, when Chinese banks were permitted to use their own judgement in setting lending rates, credit risk has been much better reflected in the lending rate setting process (People's Bank of China, 2005). Another important improvement is that lending decisions have become more centralised through a reduction of the autonomy of regional offices and branches, and of any possible interference by local and regional governments and Party officials (He and Fan, 2004; Dobson and Kashyap, 2006). All in all, it has been suggested that banks in China have been considerably freed from political pressure, for example to make loans to support state-owned enterprises (Lardy, 2006).⁹ This should foster a more competitive environment.

To conclude, this brief summary of financial reform and banking structure in China yields a number of important insights for our analysis of competitive conditions in Chinese loan markets. First, the process of financial reform has led to a fundamental change in the orientation of Chinese banks from mere instruments of economic and industrial policy

^{9.} Notwithstanding proofs of considerable government interference, Cull and Xu (2000) provide evidence that lending by Chinese banks to state-owned companies was guided by credit risk considerations as well.

towards in essence commercially oriented providers of a broad range of financial services to non-financial corporations, central, regional and local governments and households (Cheng and Degryse, 2010).¹⁰ Second, more specifically, the lending policies of Chinese banks have been founded much more on market-based criteria. This has been aided by the liberalisation of interest rates and strict credit controls. Loan rates have been liberalised gradually since 1998, first by increasing their ceilings and later by removing them altogether. Moreover, the abolishment of binding credit plans has provided banks in China with greater opportunities to diversify and optimise their lending business. As demonstrated by Abiad et al. (2010), the process of financial reform has made the biggest advances regarding credit controls. We concur with Huang (2010) that credit allocation has increasingly become more marketoriented. Third, notwithstanding the growing adherence to market principles in Chinese banks' lending policies, the fact that the PBC has continued, at least during certain episodes, to provide loan guidelines is important for our investigation. While the effectiveness of these quantitative instruments is not clear and may actually have been limited, their continued use makes us rather sceptical about employing banks' market shares to assess competitive conditions in Chinese loan markets.¹¹ Finally, the international liberalisation of China has promoted the presence and most likely the influence of foreign banks, arguably most importantly through their impact on the strategies and operations of Chinese banks. Overall, the broad range of financial reform measures implemented may have promoted over time a priori competitive conditions in Chinese banking markets.

3.2 Empirical literature on bank efficiency and competition in China

One of the most frequently investigated aspects of the Chinese banking sector is its efficiency.¹² Most studies have characterised Chinese banks as still being relatively inefficient, despite improvements since the highly regulated and inefficient banking system of the past (Fu and Heffernan, 2007; Yao et al., 2007; Berger et al., 2009a; García-Herrero et al., 2009). This also applies when they are compared with their international peers operating outside China (Singh Pritam Singh and Munisamy, 2008; Löchel and Li, 2011; Allen et al., 2012).¹³ To what extent bank efficiency in China has improved over time is still subject to considerable debate, although some studies have argued that considerable efficiency gains have been realised in more recent years (see for example: Matthews et al., 2009; Barros et al., 2011).

The discussion on efficiency improvements has concentrated on two issues, which in empirical studies often yield mixed results (Berger et al., 2009a; Chang et al., 2012): the impact of bank deregulation on bank efficiency (Kumbhakar and Wang, 2007; Fu and Heffernan, 2007) and the relationship between the efficiency and performance of banks in China. On the former, some support the view that efficiency has improved in parallel with the process of financial reform (Feyzioğlu, 2009), while others are more sceptical and tend to

^{10.} It needs to be acknowledged that some observers remain rather sceptical about the degree of progress achieved in Chinese financial reform. For example, World Bank (2012) concludes that "[d]espite the many reforms introduced so far, the Chinese financial system remains oppressed, unbalanced, costly to maintain and potentially unstable. [...] Continued protection and intervention in the business decisions of financial institutions make them convenient policy instruments, the use of which prolongs the bureaucratic culture and distorted incentives that have prevented banks from full commercialization and from allocation of financial resources to the most productive uses".

^{11.} Van Leuvensteijn et al. (2011 and 2013) use market shares in loan markets to investigate competition in a number of mature economies.

^{12.} Most of the empirical studies focus on cost and profit efficiency using the stochastic frontier approach; examples include: Fu and Heffernan (2007), Feyzioğlu (2009), Berger et al. (2009a). Some researchers focus on technology efficiency and sources of total factor productivity growth by constructing various productivity indices (Chang et al., 2012; Matthews et al., 2009). Chang et al. (2012) provide a comprehensive review of various efficiency measures applied to Chinese banking.

^{13.} In contrast, Matthews et al. (2009) argue that Chinese banks have achieved significant improvements in bank efficiency and hence may not be out of line with international bank efficiency benchmarks.

highlight that, despite tangible liberalisation efforts, China's banking sector is still relatively inefficient or not sufficiently commercially oriented (Allen et al., 2005; Dobson and Kashyap, 2006; Podpiera, 2006). A few studies document that reforms during the early 1990s did not seem to have affected performance (Park and Sehrt, 2001). On the latter, some studies demonstrate that Chinese bank performance has been affected positively by efficiency improvements, and that more profitable banks tend to be more efficient than less profitable banks (Heffernan and Fu, 2010; García-Herrero et al., 2009). In contrast, alternative ones emphasise that the high profitability of Chinese banks which has been observed in recent years, for example in terms of both return on average assets and equity in comparison with those of international peers, is not correlated with their efficiency, but is largely driven by wide interest rate spreads between loan and deposit rates and low personnel costs (Feyzioğlu, 2009; Löchel and Li, 2011).

Results of empirical studies on the efficiency of Chinese banks are the most conclusive when conducted for specific banking groups. The SOCBs seem to have been the least efficient commercial banks in China, especially in comparison with the JSCBs and/or foreign banks (Kumbhakar and Wang, 2007; Fu and Heffernan, 2007; Yao et al., 2007; Ariff and Can, 2008; Berger et al., 2009a; Feyzioğlu, 2009; Fu and Heffernan, 2009; Lin and Zhang, 2009; Matthews et al., 2009). Some argue that this may be related to their relative lack of shareholder diversification, compared with JSCBs, CCBs and foreign banks, whose plurality of shareholders may reduce political interference (Ferri, 2009). Shih et al. (2007) and Feyzioğlu (2009) find that the JSCBs perform significantly better than both the SOCBs and CCBs. Moreover, the impact of Chinese WTO membership since 2001 on bank efficiency has been investigated. Some studies show that the efficiency of domestic Chinese banks weakened after China's accession to the WTO, while that of foreign banks increased (Rezvanian et al., 2011). Regarding specific aspects of international liberalisation, a recent study shows that reforms which promoted foreign (and reduced state) ownership of Chinese banks had strong favourable effects on their efficiency (Berger et al., 2009a). García-Herrero and Santabárbara (2008) demonstrate that this applies especially to foreign participation through minority strategic partnerships (see also Hasan and Xie, 2012). Berger et al. (2010) show that foreign ownership has some positive effects on performance, as it mitigates certain adverse effects of business diversification. On the other hand, Heffernan and Fu (2010) argue that foreign equity investment did not have a significant influence on performance (measured by different indicators).

Only a few papers investigate explicitly and in-depth competition in the Chinese banking sector by using econometric tools, while others discuss it on the sidelines and often adopt more descriptive approaches. One of the first to address this issue adopts the structural approach (see Section 2) and calculates concentration indicators (HHI) which show high degrees of concentration that may inhibit competition (Wong and Wong, 2001). Furthermore, this study concentrates qualitatively on institutional characteristics which inhibited bank competition in China during the 1990s, such as government interference, information deficiencies and a weak legal infrastructure. At the same time, this study acknowledges that reforms stipulated under the conditions for China's WTO accession helped to create a more competitive and efficient banking system.

Turning to non-structural approaches, Yuan (2006), using the Panzar-Rosse method, investigates bank competition in China during 1996-2000, just before the country joined the WTO in 2001. The paper concludes that the banking system in China was already close to a state of perfect competition before foreign banks began to enter Chinese banking more

extensively. Fu and Heffernan (2009) look at the effect of financial reform on China's banking structure and performance during 1985-2002 and draw the conclusion that the estimation of structure-performance models lends some support to the existence of relative market power in the 1980s and early 1990s. Overall, they conclude that to improve competition, new policies should be adopted to encourage market entry and to increase the market share of the most efficient banks. Fu (2009) looks at competition in Chinese commercial banking, also by using the Panzar-Rosse method. Based on a sample of 76 banks for the period 1997-2006, it is concluded that China's overall banking market was perfectly competitive in 2001, but featured monopolistic competition thereafter until 2006. Thus, the paper supports the conclusion of Yuan (2006) that the Chinese banking sector was close to a state of perfect competition before China joined the WTO and that WTO membership might not promote overall bank competition further. Moreover, Fu (2009) shows that the H-statistic for core commercial banking activities was higher before WTO entry than after, suggesting that competition in Chinese loan markets was higher during the period before joining the WTO than in the period after. Bikker et al. (2007), as part of an investigation of 101 countries with the Panzar-Rosse model, also measure competition in Chinese banking and have results suggesting perfect competition. However, they warn that these results should be interpreted with great caution due to the limitations of the Panzar-Rosse model for China.

A few studies apply the Lerner index to Chinese banking. Fungáčová et al. (2012) find that competition in the Chinese banking industry declined, based on a sample of 76 banks during 2002-2011. They also find that competition differs depending on the type of banks, with foreign banks being the most competitive (ie lowest Lerner index). Also Soedarmono et al. (2013) report lower competition in Chinese banking over time, as part of an investigation of 11 Asian banking systems for 1994-2009 (for China, covering 103 banks). Both papers report Lerner indices for China that lie predominantly between 0.3 and 0.4 for 2002-2008.

All in all, the results of empirical studies, both for China and more generally (see Section 2), point to several lessons for the analysis of bank competition in China. First, it may be worthwhile to compare the results of various measures of competition, such as the Lerner index, Panzer-Rosse and PE approaches, since a priori the literature does not suggest a superior method. Second, concentration indicators do not seem to be appropriate competition measures. Hence, we do not employ them in our analysis. Finally, banking deregulation and foreign bank entry generally have affected bank competition favourably. This may be a finding especially relevant for China, taking into account its elaborate process of financial reform.

4 Competition measures: methodology and theory.

In this section, we discuss the conventional Lerner and elasticity-adjusted Lerner measures (4.1) and the relatively new Profit Elasticity (PE) indicator (4.2). We theoretically show that the former fail to measure competition correctly when interest rates are regulated, while the latter does not suffer from this shortcoming (4.3).

Our discussion of the conventional Panzar-Rosse H-statistic is conducted in Appendix A, as its flaws in measuring competition under regulated interest rates are rather clear and do not require a theoretical assessment. Namely, if due to binding regulation deposit and lending rates move in step, the H-statistic will be biased upwards and measure incorrectly a higher degree of competition. We adopt the H-statistic merely to replicate and check the results of Yuan (2006) and Fu (2009).

4.1 Lerner and elasticity-adjusted Lerner indices

4.1.1 LERNER INDEX

The Lerner index reflects firms' ability to set prices over marginal costs. Fierce competition will lower its level, as firms reduce prices towards marginal costs. In the extreme case of perfect competition, the Lerner index will be reduced to zero, while with monopoly it will reach one. The traditional Lerner index has been applied widely in empirical competition literature (Fernández de Guevara et al., 2005 and 2007; Berger et al., 2009b).¹⁴ However, to the best of our knowledge, only Fungáčová et al. (2012) conducted an in-depth analysis based on this measure for Chinese banking markets during the post-WTO accession period. Our approach differs in the sense that we do not focus on bank competition in general but instead concentrate on competition in loan markets. Hence, we define the Lerner index as:

$$L_{it} = \left(p_{it} - mc_{it}\right) / p_{it} \tag{4.1}$$

where p_{it} denotes the price of a loan for bank *i* at time *t*, which is defined as total interest income divided by total loans, while mc_{it} are marginal costs of loans.

In order to be able to calculate marginal costs of loans, we first estimate a Translog Cost Function (TCF) using individual bank observations (Van Leuvensteijn et al., 2007). This function assumes that the technology of an individual bank can be described by one multiproduct production function. Under proper conditions, a dual cost function can be derived from such a production function, using output levels and factor prices as arguments. A TCF is a second-order Taylor expansion around the mean of a generic dual cost function with all variables appearing as logarithms. It is a flexible functional form that has proven to be an effective tool in explaining multiproduct bank services. Our TCF has the following form:

$$\ln c_{it} = \alpha_0 + \sum_{t=1,\dots,(T-1)} \gamma_t d_t + \sum_{h=1,\dots,(H-1)} \eta_h d_h$$

+
$$\sum_{j=1,\dots,K} \delta_j \ln x_{ijt} + \sum_{j=1,\dots,K} \sum_{k=1,\dots,K} \epsilon_{jk} \ln x_{ijt} \ln x_{ikt} + v_{it}$$
(4.2)

where the dependent variable c_{it} reflects the production costs of bank i (i = 1, ..., N) in year t (t = 1, ..., T). d_t are year dummies and d_h are bank type dummies (h = SOCB, JSCB,

^{14.} The various measures that we adopt in this paper can be linked theoretically. For example, Shaffer (1983) demonstrated that the Lerner index can be derived in terms of the Panzar-Rosse H-statistic. We consider this as future work.

CCB).¹⁵ The explanatory variables x_{ikt} represent three groups of variables (k = 1, ..., K). The first group consists of (K_1) bank output components, such as loans, securities and other services (proxied by other income). The second group consists of (K_2) input prices, such as wage rates, deposit rates (as price of funding) and the price of other expenses (proxied as the ratio of other expenses to fixed assets). The third group consists of ($K - K_1 - K_2$) control variables, eg the equity ratio. In line with Berger and Mester (1997), the equity ratio corrects for differences in loan portfolio risk across banks (Van Leuvensteijn et al., 2007). v_{it} is the error term.

Two standard properties of cost functions are linear homogeneity in the input prices and cost exhaustion (see eg Beattie and Taylor, 1985; Jorgenson, 1986). They impose the following restrictions on the parameters, assuming – without loss of generality – that the indices *j* and *k* of the two sum terms in equation (4.2) are equal to 1, 2 or 3, respectively, for wages, funding rates and prices of other expenses:

$$\delta_1 + \delta_2 + \delta_3 = 1, \epsilon_{1j} + \epsilon_{2jk} + \epsilon_{3j} = 0 \text{ for } j = 1, 2, 3, \text{ and } \epsilon_{k,1} + \epsilon_{k,2} + \epsilon_{k,3} = 0 \text{ for } k = 4, ..., K$$
(4.3)

The first restriction stems from cost exhaustion, reflecting the fact that the sum of cost shares is equal to unity. In other words, the value of the three inputs is equal to total costs. Linear homogeneity in the input prices requires that the three linear input price elasticities (δ_i) add up to one, whereas the squared and cross terms of all explanatory variables (ϵ_{ij}) add up to zero. Again without loss of generality, we also apply symmetry restrictions $\epsilon_{j,k} = \epsilon_{jk}$ for j, k = 1, ..., K.

The marginal costs of loans are obtained by differentiating the TCF (4.2) with respect to loans, namely:

$$mc_{i1t} = c_{it} / x_{i1t} \left(\delta_1 + 2\epsilon_{1t} \ln x_{itt} + \sum_{k=1,\dots,K:k\neq l} \epsilon_{1k} \ln x_{ikt} \right)$$

$$\tag{4.4}$$

Once marginal costs of loans are obtained, an individual bank's Lerner index is calculated according to equation (4.1). The yearly Lerner index L_t is then the average of the individual L_{it} for each year t, and the subsample Lerner index $L_{subsample}$ is the average of the individual bank's Lerner indices for each subsample. The subsamples are the periods pre-WTO (1996-2001) and post-WTO (2002-2008).

4.1.2 ELASTICITY-ADJUSTED LERNER INDEX

The traditional Lerner index cannot distinguish markets that have high margins due to inelastic demand from markets that have high margins because they are less competitive or perhaps collusive (Corts, 1999, p. 31). To overcome this problem, the elasticity-adjusted Lerner index has been developed (Genesove and Mullin, 1998; Corts, 1999; Wolfram, 1999; Van Leuvensteijn, 2008). More precisely, this measure normalises the Lerner index for the price elasticity of demand.

We estimate the elasticity-adjusted Lerner index following Angelini and Cetorelli (2003). We provide only a very brief introduction of this indicator, since it is rather standard in the literature. Bank *i* solves the following profit-maximising problem:

^{15.} In this section, we assume that cost functions for each bank type are similar, as only the constant term is allowed to vary across bank groups through bank type dummies. The alternative approach is to assume different cost functions for each bank type by allowing bank type dummies to interact with independent variables. We follow this approach in Appendix E (E.5) as an additional robustness test.

$$Max_{qi}\Pi = p(Q)q_i - C(q_i, w_i)$$

$$\tag{4.5}$$

where $Q = \Sigma q_i$ is the total amount of bank loans in loan markets and q_i is the loan provided by bank *i*. $C(q_i, w_i)$ is the cost function of bank *i*, and w_i represents the vector of factor input prices. The corresponding first-order condition is:

$$\boldsymbol{p}_{i} = \mathsf{C}'(\boldsymbol{q}_{i}, \boldsymbol{w}_{i}) - \boldsymbol{\Theta}_{i}/\boldsymbol{\varepsilon}$$

$$\tag{4.6}$$

where Θ_i is defined as the conjectural elasticity of total industry output with respect to the output of bank *i*, and ε is the market semi-price elasticity of demand, namely $\Theta_i = (dQ/dq_i)/(Q/q_i)$ and $\varepsilon = (dQ/dp)/Q$. In a perfectly competitive market, Θ_i equals zero for all banks *i*, while in a monopoly market Θ_i equals one. The separate identification of these two elasticities requires the simultaneous estimation of a supply and demand equation (Angelini and Cetorelli, 2003).

Appelbaum (1982) suggests that it is sufficient to estimate the ratio $\lambda = \Theta/\varepsilon$ if the goal is to evaluate the industry's overall degree of market power.¹⁶ The elasticity-adjusted Lerner index will then be defined as $L = \lambda/p$, where *p* is the average price of loans. Market power depends on both the elasticity of demand and the degree of competition, measured by conjectural variation.

To identify λ and the elasticity-adjusted Lerner index, we estimate simultaneously a Translog Cost Function (TCF) and the supply equation, imposing cross-equation restrictions. The TCF and marginal costs of loans are defined the same way as equations (4.2) and (4.4), respectively.

Substituting the marginal costs equation (4.4) into the supply equation (4.6), we obtain:

$$p_{it} = \frac{c_{it}}{x_{ilt}} \left(\delta_1 + 2\epsilon_{1l} ln x_{ilt} + \sum_{k=1\dots K; k \neq l} \epsilon_{1k} ln x_{ikt} \right) + \sum_{t=1\dots T-1} \lambda_t d_t + \varepsilon_{it}$$
(4.7)

where d_t is a year dummy and ε_{it} is the error term.

To access the evolution of bank competition measured by the elasticity-adjusted Lerner index, we perform two types of regressions: yearly estimates and subsample estimates. The yearly elasticity-adjusted Lerner index is then derived as λ_t/p_t , where p_t is the yearly average loan rate. To obtain subsample estimates of the elasticity-adjusted Lerner index, we regress simultaneously equations (4.2) and (4.7), replacing year dummies d_t with subsample dummies in both equations. The subsample elasticity-adjusted Lerner index is then defined as $\lambda_{subsample}/p_{subsample}$, where $p_{subsample}$ is the average loan rate for each subsample. The estimation is carried out with three-stage least squares (3SLS). To control for endogeneity of the cost and quantity variables, we employ one-period lagged variables as instruments; therefore the results are available starting from 1997.

Unlike the traditional Lerner index, the elasticity-adjusted Lerner index has the advantage of allowing for formal tests whether its estimated values over time are statistically different from zero or one and whether subsample estimates differ significantly. These

^{16.} As a robustness test, we estimate explicitly the conjectural variation parameter as a direct measure of competition in Appendix E (E.3).

advantages are crucial for our research, as we want to investigate if competitive conditions changed significantly over time.

4.2 The Profit Elasticity (PE) model

The PE indicator, or Relative Profit Differences (RPD), is based on the notion, first, that more efficient firms (that is, firms with lower marginal costs) gain higher market shares or profits and, second, that this effect is stronger the heavier the competition in that market is (Van Leuvensteijn et al., 2007). While RPD can be seen as the theoretical model underlying this work, the PE indicator is the empirical operationalisation of this model. Boone (2008) shows that there is a continuous and monotonically increasing relationship between RPD and the level of competition if firms are ranked by decreasing efficiency. In other words, there is a negative relationship between efficiency, measured in terms of marginal costs, and profits; the more intense this negative relationship is, the more competitive markets will be. So, in practice, the PE indicator will have a negative sign when the relationship between marginal costs and profits is estimated, and it will be more negative the higher the level of competition is.

The fact that this relationship is both continuous and monotonic is the main advantage of RPD over more traditional measures of competition such as the HHI and Lerner index (or PCM approaches). Another advantage is that RPD and the PE indicator are not dependent on assumptions about the type of competitive model, such as whether this is Bertrand or Cournot competition.¹⁷

The PE indicator (also referred to as the Boone indicator) has been developed in a broad set of theoretical models (Boone, 2000, 2001 and 2008; Boone et al., 2004; Boone et al., 2007; CPB, 2000). Following Boone et al. (2004), and replacing "firms" with "banks", we consider a banking industry where each bank *i* produces one product q_i (or portfolio of banking products), which faces a demand curve of the form:

$$p(q_i, q_{j\neq i}) = a - bq_i - d\Sigma_{j\neq i}q_j$$
(4.8)

and has constant marginal costs mc_i . This bank maximises profits $\pi_i = (p_i - mc_i) q_i$ by choosing the optimal output level q_i . We assume that $a > mc_i$ and $0 < d \le b$. The first-order condition for a Cournot-Nash equilibrium can then be written as:

$$a - 2bq_i - d\Sigma_{i\neq j}q_j - mc_i = 0 \tag{4.9}$$

When N banks produce positive output levels, we can solve the N first-order conditions (4.9), yielding:

$$q_{i}(mc_{i}) = \frac{\left[\left(2b/d - 1\right)a - \left(2b/d + N - 1\right)mc_{i} + \Sigma_{j}mc_{j}\right]}{\left[\left(2b + d(N - 1)\right)\left(2b/d - 1\right)\right]}$$
(4.10)

We define profits π_i as variable profits excluding entry costs ε . Hence, a bank enters the banking industry if, and only if, $\pi_i \ge \varepsilon$ in equilibrium. Note that Equation (4.10) provides a relationship between output and marginal costs.

^{17.} Cournot competition is an economic model used to describe an industry structure in which companies compete on the amount of output they will produce, which they decide on independently of each other and at the same time. Bertrand competition assumes that firms compete on price and not output quantity (market shares). For more detail, see Tirole (1988).

It follows from π_i (*mc_i*) = (p_i - *mc_i*) q_i that profits depend on marginal costs in a quadratic way, ie

$$\pi_i(mc_i) = \frac{(2b/d-1)a - (2b/d+N-1)mc_i + \sum_j mc_j}{[(2b+d(N-1))(2b/d-1)]} (p_i - mc_i)$$
(4.11)

The theoretical concept RPD is then defined as $(\pi(mc^{-}) - \pi(mc))/(\pi (mc^{-}) - \pi(mc)))$ for any three firms with $mc^{-} < mc^{-} < mc^{-}$. In this market, competition can increase in two ways. First, competition increases when the produced services of the various banks become closer substitutes, that is, *d* increases (keeping *d* below *b*). Second, competition increases when entry costs ε decline. Boone (2008) proves that RPD is an increasing function of interaction among existing firms (dRPD/dd>) and a decreasing function of entry costs ($dRPD/d\varepsilon$ <0). In other words, RPD increases when competition intensifies, ie fiercer competition increases (decreases) profits of more efficient firms by larger (smaller) amounts than those of less efficient firms. Hence, competition rewards efficiency, a concept that can be traced back to Demsetz's (1973) efficiency structure hypothesis.

Boone (2008) demonstrates how RPD can measure the level and evolution of competition in practice. Firms are first ranked by their efficiency level. Subsequently, RPD of firm *i* are normalised by calculating its relative profit difference against the profits of the most and the least efficient firms. This procedure yields a normalised RPD curve as a function of normalised relative efficiency differences. The level of competition is then represented by the area under the normalised RPD curve. Since changes in competition move all points on the RPD curve monotonically, shifts in this curve measure the evolution in competition.

Although this procedure is mathematically elegant, it is computationally intensive, as it requires the ranking of firms by efficiency levels (ie marginal costs) for each year. Conversely, most empirical studies that adopt Boone's work regress the logarithm of profits on the logarithm of marginal costs to capture the essence of RPD. They refer to the estimated elasticity of profits with regard to marginal costs, ie $dln(\pi)/dln(mc)$, as the PE indicator (Boone et al., 2004 and 2007; Van Leuvensteijn et al., 2007; Tabak et al., 2012). This indicator is in theory negative, reflecting the fact that higher marginal costs are associated with lower profits. In addition, its value should be lower the more competitive market conditions are. The PE indicator is based on the same theoretical foundation as RPD, as they both capture the central idea that less efficient firms are punished more in more competitive markets. Boone et al. (2007) conducted simulations for the PE indicator and found that changes in competition are correctly identified with this measure. Unlike the computationally intensive RPD, the PE indicator has the advantage that it can be easily estimated in practice and has a rather straightforward interpretation. We therefore employ the PE indicator to measure competition in the next section.

We note that the PE indicator model, like every other model, is a simplification of reality (Van Leuvensteijn et al., 2007). First, efficient banks may choose to translate lower costs either into higher profits or into lower output prices in order to gain market share. Our approach assumes that banks in China compete on efficiency in order to predominantly increase profits and not to expand market share, given quantitative restrictions in the form of explicit lending quotas and informal window guidance (see Section 3.1). Even when some banks would choose to increase profits by lowering their price and increasing their market share, the PE indicator would also measure this effect. Still, we assume that this

ignores differences in bank product quality and design, as well as the attractiveness of innovations. We assume that banks are forced over time to provide quality levels that are more or less similar. By the same token, we presume that banks have to follow the innovations of their peers. Hence, like many other model-based measures, the Boone indicator approach focuses on one important relationship (that between efficiency and profits), thereby disregarding other aspects (see also Bikker and Bos, 2005). All in all, the PE indicator may be applied in relatively homogeneous product markets where product innovation and differences in quality do not matter too much. Therefore, we focus only on competition in loan markets and not on overall bank competition in China. Naturally, annual estimates of the PE indicator are more likely to be impaired by these distortions than the estimates covering the full sample period. Hence, in addition to annual estimates, we provide point estimates for the full 1996-2008 period and the subsample periods 1996-2001 and 2002-2008, 2001 being the year of China's entry into the WTO.

4.3 Competition measures under interest rate regulation

To understand the direct effect of binding deposit rate regulation on the Lerner index and RPD, we consider the simple model described in Section 4.2. Binding deposit rate regulation in China affects the level of marginal costs of all banks and redistributes market share between efficient and inefficient banks. We show below that this redistribution of output can result in both increasing and decreasing competition as indicated by the Lerner index, which makes it an inconsistent measure of competition under binding deposit rate regulation. On the other hand, RPD is continuous and monotone in competition in a market with binding deposit rate regulation. In the following exercise, we assume that the slope of the loan demand function does not change after exogenous movements in input prices. To keep it simple, we also assume that deposit rate regulation does not allow market exit and entry due to changes in deposit rate regulation.

Imposing a deposit rate ceiling should reduce the level of competition because more efficient banks cannot undercut less efficient rivals by setting deposit rates above the ceiling. Less efficient banks then are protected by the ceiling and are less likely to be forced out of the market. Abolishing or raising deposit ceilings should increase competition because more efficient banks can expand market share at the expense of their less efficient rivals.

We assume that deposit rate regulation has a homogeneous impact on each bank's marginal costs. Then, under regulation, a bank's marginal cost of loans becomes $mc_i(\varepsilon) = mc_i - \varepsilon$ (i = 1, ..., N). ε is a regulation parameter, which measures the extent to which deposit rate regulation is binding. $\varepsilon \in (-\varepsilon, mc)$, where mc is the marginal cost of the most efficient bank and ε is some positive number that allows the least efficient bank to remain profitable and stay in the market. A positive ε reflects a binding deposit rate ceiling, while a negative ε corresponds to a binding deposit rate floor. Higher values of ε lead to less competition. This parameter can be time-variant to reflect changes in regulation across time. We focus here on deposit rate regulation. Nevertheless, this general setup can also be applied to other regulations (or technology shocks) that impact homogenously upon the cost side of banks. From equations (4.8), (4.9) and (4.10), and imposing ε , we derive the effect of binding deposit rate regulation on optimal output:

$$f(\varepsilon) = q_i(\varepsilon) - q_i$$

$$= \frac{\left(\frac{2b}{d} - 1\right)\varepsilon}{\left(2b + d\left(N - 1\right)\right)\left(\frac{2b}{d} - 1\right)}$$
(4.12)

where q_i is the optimal output without deposit rate regulation. Given $0 < d \le b$, $f(\varepsilon)$ is increasing in ε and takes the same sign as ε . Hence, under a deposit rate ceiling (floor), each bank's optimal output increases (decreases) by the same amount. We write the Lerner index for bank *i* as a function of regulation-free optimal output, marginal costs and the regulation parameter ε :

$$L_{i}(\varepsilon) = \frac{bq_{i}(\varepsilon)}{\left(bq_{i}(\varepsilon) + mc_{i}(\varepsilon)\right)}$$

$$= \frac{b\left(q_{i} + f(\varepsilon)\right)}{\left(b\left(q_{i} + f(\varepsilon)\right) + mc_{i} - \varepsilon\right)}$$
(4.13)

Taking the derivative with respect to $\boldsymbol{\varepsilon}$ and using $f'(\boldsymbol{\varepsilon}) = f(\boldsymbol{\varepsilon})/\boldsymbol{\varepsilon}$, we obtain:

$$sign\left(\frac{dL_{i}(\varepsilon)}{d\varepsilon}\right) = sign\left(b\left(q_{i} + \frac{f(\varepsilon)}{\varepsilon}mc_{i}\right)\right)$$

$$> 0$$
(4.14)

Hence, a higher value of ε increases an individual bank's Lerner index, indicating less competition, as theory would suggest. However, the aggregate Lerner index – ie for the whole market – might not give a consistent value because the market shares of efficient banks decrease due to deposit rate regulation. To see this, define the market share of bank *i* as s_i (ε) = $q_i(\varepsilon)/\Sigma_i q_i$ (ε), and define bank *k* as the bank that produces at market average marginal costs, namely $mc_k = \Sigma_j mc_i/N$. Market share under deposit rate regulation can then be written as:

$$s_{i}(\varepsilon) = \frac{1}{N} \frac{\left(\frac{2b}{d} - 1\right)a - \left(\frac{2b}{d} + N - 1\right)mc_{i} + \Sigma_{j}mc_{j} + \left(\frac{2b}{d} - 1\right)\varepsilon}{\left(\frac{2b}{d} - 1\right)a - \left(\frac{2b}{d} + N - 1\right)mc_{k} + \Sigma_{j}mc_{j} + \left(\frac{2b}{d} - 1\right)\varepsilon}$$
(4.15)

Taking the derivative with respect to ε yields:

$$sign\left(\frac{ds_{i}}{d\varepsilon}\right) = sign\left(\left(\frac{2b}{d} - 1\right)\left(\frac{2b}{d} + N - 1\right)\left(mc_{i} - mc_{k}\right)\right)$$
(4.16)

It is immediately clear that the market share of bank *i* increases with a higher ε if, and only if, $m_{Ci}>m_{Ck}$. Therefore, regulation reallocates market share from efficient banks to less efficient banks (eg banks with marginal costs above the market average). The effect of binding deposit rate regulation on the aggregate Lerner index is then:

$$\frac{dL}{d\varepsilon} = \sum_{i=1}^{k} \frac{ds_i}{d\varepsilon} L_i + \sum_{i=k+1}^{N} \frac{ds_i}{d\varepsilon} L_i + \sum_{i=1}^{N} s_i \frac{dL_i}{d\varepsilon}$$
(4.17)

Denote banks i = 1,..., k as low-efficiency banks, which will see their market shares increase. In contrast, the market share of high-efficiency banks i = k+1,..., N will decrease. All in all, this leaves the sign of $dL/d\varepsilon$ undetermined. Specifically, if deposit rate regulation reallocates sufficient market share from efficient to less efficient banks (resulting in $dL/d\varepsilon$ <0), then competition such as measured by the Lerner index can increase instead of decrease. This simple example shows that the aggregate Lerner index cannot consistently measure competition under deposit rate regulation.¹⁸

In contrast, RPD is not biased due to interest rate regulation. As described in Section 4.2, RPD is defined as the ratio of the profit differences between any three banks in the market. Banks can be ordered by their efficiency level (marginal costs), with more efficient banks providing more loans. Suppose we take three banks – A, B, C – with $m_{CA} < m_{CB} < m_{CC}$, then RPD is defined as $RPD = (\pi_A - \pi_C)/(\pi_B - \pi_C)$. Using the model presented in Section 4.2, profits can be written as a quadratic function of outputs. Then, after imposing deposit rate regulation, $RPD(\varepsilon)$ can be rewritten as:

$$RPD(\varepsilon) = \frac{(q_{A} + f(\varepsilon))^{2} - (q_{C} + f(\varepsilon))^{2}}{(q_{B} + f(\varepsilon))^{2} - (q_{C} + f(\varepsilon))^{2}}$$
(4.18)

We show that RPD (ϵ) is decreasing in ϵ by taking the first-order derivative:

$$sign\left(\frac{dRPD}{d\varepsilon}\right) = sign\left(\frac{2f'(\varepsilon)(q_{B} - q_{A})}{\left(q_{B} + q_{c} + 2f(\varepsilon)\right)^{2}}\right)$$

$$< 0$$
(4.19)

Given that $q_B - q_A < 0$ and $f'(\varepsilon) > 0$, the above equation has a negative sign, suggesting that higher binding regulation (ie a higher ε) will lower competition, consistent with theory. Hence, RPD is a consistent measure of competition in case of binding deposit rate regulation.

We show below the two main problems with the (elasticity-adjusted) Lerner index when lending rate regulation is binding. First, this index mainly measures variation in competition resulting from changing regulation; it cannot detect competition resulting from shifts in demand. Second, ignoring binding price regulation leads to inconsistent estimates of the elasticity-adjusted Lerner index (see also Salvo, 2010).

Consider the simple case of a monopoly bank serving the entire market under a lending rate ceiling.¹⁹ If this ceiling is not binding (see Panel A of Figure 2), the bank will choose the optimal price and quantity combination by equating marginal cost (*MC*) to marginal revenue (*MR*). When the demand curve shifts from *D1* to *D2* (da > 0),²⁰ the equilibrium combination of prices and output moves from point *E1* to *E2*, resulting in a higher Lerner index, or lower competition. Hence, changes in competition resulting from

^{18.} Boone (2000) provides another example where an individual firm's Lerner index increases after competition intensified. Applying that model with a slight modification, it can be shown that the necessary condition for an individual bank's Lerner index to be increasing in ε is that the marginal cost of this bank is lower than the market average. Proof is available upon request.

^{19.} Competition is a concept closely related to market power, and in most of the literature they are considered in a similar fashion. Even for a monopoly, the issue of market power is relevant. We use a monopoly here for reasons of simplicity. The example is also valid for a market with multiple firms. See Koetter et al. (2008) for more details. **20.** For a full proof that da > 0 leads to lower competition, please refer to Boone (2000).

exogenous shifts in the demand curve can be correctly picked up by the Lerner index. However, this is not the case if the lending rate ceiling is binding, as demonstrated in Panel B. This ceiling (P_c) prevents the bank from choosing the optimal price-output combination according to MR = MC. In contrast, following profit-maximising behaviour, it will choose the quantity at the kink of the demand curve (points *E1* and *E2* of Panel B), leaving the price unchanged at the ceiling. Therefore, changes in competition due to exogenous shifts in demand cannot be indentified by the Lerner index, because both prices and costs do not change in relation to the change in demand.

In case both the demand curve and binding lending rate ceiling change, the Lerner index can pick up only variations in competition due to changes in the latter, but not those due to shifts in the former. In Panel B, suppose that the demand curve shifts to D2 and the lending rate ceiling increases to P_c ', both of which will decrease competition. The optimal combination of prices and output moves from E1 to E2' and hence the Lerner index increases. It is immediately clear that changes in this index reflect only changes in the lending rate ceiling but not in the demand curve, because the new Lerner indices are the same with or without demand curve shifts (comparing E2' and E1'). All in all, in the case of a binding lending rate ceiling, the Lerner index provides only an incomplete assessment of changes in competition.

The above analysis also applies to the elasticity-adjusted Lerner index because it estimates the price-cost margin of an average bank. This conclusion is closely related to the analysis in Salvo (2010), which proves theoretically and empirically that ignoring price ceilings result in an over-estimation of competition by the elasticity-adjusted Lerner index in the context of the Brazilian cement industry. When prices are unconstrained, the traditional joint estimation approach (eg Bresnahan, 1982) can effectively distinguish between monopoly and perfect competition. In contrast, when prices are regulated (for example, a price ceiling is put in place), demand shifts do not affect prices in the case of both a monopoly and perfect competition.



Thus, unless marginal costs are observed, one cannot tell whether the observed price-quantity combination is established under a monopoly or perfect competition. If one were to ignore the existence of a price ceiling and hence conclude that prices remain stable after a shift in demand, one would falsely reject collusion and argue in favour of competition. In general, if binding price ceilings are not properly accounted for, the underlying structural model will be misspecified. Hence, the orthogonality condition that is required for a consistent estimation of the related parameter will not be met. Salvo (2010) further shows that ignoring price ceilings may lead to an over-estimation of competition, in line with our argumentation.

Overall, we conclude that the (elasticity-adjusted) Lerner index is a biased measure of competition when price ceilings are binding. We suspect that this may account for the very high level of competition that it obtains for the pre-WTO period in China. It is generally acknowledged that the lending rate ceiling was most likely binding during this period.

In contrast, RPD uses relative profits and therefore they can pick up changes in competition due to demand shifts under price ceilings. For illustrative purposes, we use a simplifying assumption for the additional demand that may result from a binding price ceiling. Specifically, we assume that the extra output will be shared among banks according to their market share without the price ceiling. This so-called repartition rule relates to Schmalensee (1987). It should be noted that our proof does not depend on any specific repartition rule, as long as it allows more efficient banks to take on relatively more additional output after the price ceiling is imposed. For simplicity, we assume that b=d, meaning that the products provided by different banks are perfect substitutes. Denote aggregate loans that are provided under the price ceiling as $Q^* = \frac{a - \overline{P}}{b}$; without the ceiling, it is Q. If the ceiling is binding, $Q' \ge Q$. Moreover, banks share the additional output $Q^* - Q$ according to their original market share s_i when there is no price ceiling. Then the optimal output for bank *i* is:

$$q_{i}^{*} = q_{i} + s_{i} \left(Q^{*} - Q \right) = s_{i} Q^{*}, \text{ where } s_{i} = \frac{1}{N} \frac{a - (N+1)mc_{i} + \Sigma_{j}mc_{j}}{a - (N+1)mc_{k} + \Sigma_{j}mc_{j}}.$$
(4.20)

Again mc_k are the marginal costs of producing loans for an average bank. Then profits of bank *i* are $\pi_i^* = (\overline{P} - mc_i)s_iQ^*$. We focus on the demand shift parameter *a* and prove that an increasing *a* leads to lower competition under the price ceiling when measured by RPD. We reiterate that in this case the Lerner index would not detect any changes in competition. The RPD of any three banks under price ceiling is:

$$RPD(a) = \frac{\left(\overline{P} - mc_{A}\right)s_{A} - \left(\overline{P} - mc_{C}\right)S_{c}}{\left(\overline{P} - mc_{B}\right)s_{B} - \left(\overline{P} - mc_{C}\right)S_{c}}$$
(4.21)

Taking the derivative with respect to a, and using $m_{CA} < m_{CB} < m_{CC}$, it can be shown that:

$$sign\left(\frac{dRPD}{da}\right) = sign\left(\left(N+1\right)\left(mc_{c}-mc_{B}\right)\left(mc_{A}-mc_{C}\right)\left(mc_{B}-mc_{A}\right)\right)$$

$$<0$$
(4.22)

Hence, RPD correctly picks up changes in competition due to demand shifts when a price ceiling is put in place. This is its main advantage (and of the PE indicator as well) when

compared with the Lerner index. Both RPD and the PE indicator can measure competition correctly under price ceilings, while the Lerner index can only measure changes in competition resulting from changed ceilings, but not those resulting from shifts in demand.

Finally, the existence of binding interest rate regulations can exarcerbate other shortcomings of conventional competition measures such as the Lerner index. A case in point is the reallocation effect identified in Boone et al. (2007). This relates to the fact that more intensive competition due to more aggressive conduct will reallocate output and profits from less efficient banks to more efficient banks. As more efficient banks usually have higher PCMs than less efficient banks, the PCM for the whole market, which is an (output) weighted average of individual banks' PCMs, actually may increase in response to more intense competition. The increase in the market PCM (or aggregate Lerner index) would be interpreted as a decline in competition, while actually it has increased. Boone et al. (2007) show that the reallocation effect is particularly strong in concentrated markets. As a matter of fact, Chinese loan markets are highly concentrated markets, where during 2001-2008 the four SOCBs had an average annual market share of around 71%. It can be demonstrated that the reallocation effect is more profound when the regulation of interest rates is binding. Hence, this should make the Lerner index even less appropriate as an indicator to measure competition in Chinese loan markets.

5 Data and sample

The main data source of our analysis is BankScope. We collect Chinese banks' balance sheet data running from 1996 to 2008. This 13-year period is selected under consideration of data availability and to capture various banking sector reforms, including those related to WTO accession. Whenever BankScope does not provide sufficient information, we use various issues of the Almanac of China's Finance and Banking, China Statistical Yearbook and individual banks' annual reports to double-check and fill in missing data.

We focus on state-owned commercial banks (SOCBs), joint-stock commercial banks (JSCBs), city commercial banks (CCBs) and foreign banks (FOREIGN). There are other types of financial institutions in China, such as trust and investment corporations, rural commercial banks, savings banks, co-operative banks, investment banks and policy banks. We exclude these institutions from our investigation for several reasons. First, in the 1990s, trust and investment corporations were important financial institutions that operated similarly to commercial banks, but with less restrictions and regulations (Hong and Yan, 1997). However, in the late 1990s, they experienced significant problems and most of them were taken over by commercial banks. Since the primary focus of this paper is to assess bank competition during 1996-2008, we believe it is safe to exclude trust and investment corporations from our analysis. Second, most of the other banks that are not included in our investigation capture only very small portions of Chinese lending markets and/or were established with different objectives from commercial banks. Third, there are significant data limitations for especially the large number of small banks that are excluded from the sample.

In order to exclude irrelevant and unreliable observations, banks are incorporated in our sample only if they fulfilled the condition that total assets, loans, deposits, equity and other non-interest income should be positive. We lost 43 observations after applying this criterion, mainly due to negative non-interest income (34 observations). At the end, we are left with 714 observations covering 1996-2008. Our sample includes extensive information on 127 banks, including all four SOCBs, all 13 JSCBs, 28 foreign banks²¹ and 82 CCBs. Table 2 summarises the distribution of the observations. Table 3 gives a short description of the variables used in the estimations, such as costs, loans, securities and other services, each expressed as a share of total assets, income or funding. Costs are defined as the sum of interest expenses, personnel expenses and other expenses.

^{21.} Banks with more than 50% foreign ownership are classified as foreign banks. We only include foreign banks that provide separate balance sheet data for the People's Republic of China. This means that several banks headquartered in Hong Kong SAR and which are classified as foreign banks by the CBRC but do not provide separate balance sheet data for their operations in mainland China are excluded from our sample.

Number and bank distribution of observations Table 2											
	SOCB	JSCB	ССВ	FOREIGN	Observations						
1996	4	9	1	4	18						
1997	4	10	3	6	23						
1998	4	10	5	7	26						
1999	4	10	9	7	30						
2000	4	10	14	5	33						
2001	4	10	17	7	38						
2002	4	10	27	8	49						
2003	4	10	33	8	55						
2004	4	12	40	8	64						
2005	4	12	55	10	81						
2006	4	13	74	11	102						
2007	4	13	73	26	116						
2008	4	13	36	26	79						
Total observations	52	142	387	133	714						
Number of banks	4	13	82	28	127						

Mean values of key variables by bank group

Table 3

Panel A																									
				FOREIG	i				FOREIG					FOREIC	3				FOREIG	i				FOREIG	
	SOCB	JSCB	CCB	Ν	WHOLE	SOCB	JSCB	CCB	Ν	WHOLE	SOCB	JSCB	CCB	Ν	WHOLE	SOCB	JSCB	CCB	Ν	WHOLE	SOCB	JSCB	CCB	Ν	WHOLE
																					Inter	est expe	nses as a	share of	total
	Tota	al costs a	s a share	e of total	assets	Lo	bans as a	share of	total ass	ets	Sec	urities as	a share o	of total a	assets	Other	services	as a share	e of total	income			funding		
Average 1996- 2001 Average 2002-	6.57	4.4	3.99	4.05	4.52	60.28	51.19	48.28	47.77	50.91	12.38	15.29	21.78	6.32	16.06	3.96	5.92	13.43	7.46	8.32	5.6	3.08	2.94	4.85	3.75
2008 Average 1996-	2.92	3.22	3.51	2.83	3.31	55.43	58.76	54.11	57.85	55.55	28.78	21.77	22.96	10.55	21.4	9.26	6.08	9.47	18.15	10.3	1.52	1.83	2.15	2.12	2.06
2008	4.61	3.71	3.57	3.16	3.6	57.66	55.61	53.38	55.12	54.46	21.21	19.08	22.81	9.81	20.22	7.26	6.01	9.98	15.4	9.85	3.4	2.35	2.25	2.78	2.45
Panel B																									
																Perso	nnel exp	enses as	a share (of total					
	Other	expenses	s as a sha	are of fixe	ed assets	Ave	erage ma	irket shar	e of lend	ling ¹	Interes	t income	as a sha	re of tot	al assets			assets ²			Interes	t income	e as a sha	re of tota	al loans
Average 1996- 2001 Average 2002-	33.8	69.12	58.18	108.77	68.04	22.01	1.07	0.16	0.02	3.57	6.95	5.23	4.4	6.14	5.41	0.5	0.4	0.63	0.57	0.52	11.64	10.44	9.54	16.36	11.47
2008 Average 1996-	40.04	68.95	56.22	261.46	93.66	17.83	1.89	0.11	0.06	1.28	3.69	3.85	4.35	3.51	4.09	0.57	0.41	0.49	0.59	0.5	6.86	6.65	8.07	8.46	7.86
2008	37.16	69.02	56.47	224.22	87.68	19.76	1.55	0.12	0.05	1.82	5.19	4.43	4.36	4.14	4.39	0.54	0.41	0.51	0.59	0.5	9.07	8.22	8.26	10.36	8.69
Panel C																									
	Othor			ate to tot	al acasta																				

	Other I	ther non-earning assets to total assets																					
		ratio ³					Fu	nding m	ix ⁴			Equity to	total as	sets ratio		Other	income	to interes	st income	e ratio			
Average 1996-																							
2001	6.87	4.83	6.91	4.03	5.55	92.03	91.14	90.41	97.14	91.75	4.54	5.64	5.78	36.43	12.17	2.04	6.77	17.36	7.30	9.32			
Average 2002-																							
2008	0.24	1.15	2.63	2.55	2.27	96.08	91.59	89.05	58.76	84.93	3.04	4.10	5.02	26.93	8.67	10.65	6.47	11.00	16.66	11.28			
Average 1996-																							
2008	3.23	2.68	3.18	2.95	3.04	94.30	91.41	89.22	64.96	86.36	3.72	4.74	5.11	29.50	9.49	6.77	6.60	11.80	14.41	10.83			

All data are in percentages. "WHOLE" represents the figure for all banks in the sample.

¹ This represents the average market share of total loans (by all banks) per individual SOCB, JSCB, CCB or foreign bank. ² Personnel expenses to assets ratio serves as a proxy of wage rate. ³ Other nonearning assets to total assets ratio is defined as: (total assets minus loans minus other earning assets)/total assets. ⁴ The funding mix is defined as: customer deposits/(total funding minus deposits from banks).

Source: BankScope, authors' own calculations.

We divide the whole sample into two subsamples, ie the pre-WTO era (1996-2001) and post-WTO era (2002-2008). We choose 2001 as the break year because China's WTO accession at the end of 2001 could potentially have affected the operation of Chinese banks, as it was accompanied by important financial reforms, and therefore may fundamentally have changed the competitive structure of Chinese banking markets. On average, total costs as a share of total assets were significantly lower for 2002-2008 for all types of banks (but the most for SOCBs), which is mostly due to lower interest expenses driven by lower deposit rates in the post-WTO era.

The wage rate, which is proxied by the ratio of personnel expenses to total assets, did not change significantly. The ratio of other expenses to fixed assets stayed relatively stable for Chinese banks as well, while it increased significantly in the post-WTO era for foreign banks.

Turning to revenues, the ratio of interest income to total assets decreased on average significantly for all types of banks, which is most likely the result of lower lending rates in the post-WTO era and increasing diversification of business activities. The latter is reflected in the development of the ratio of other services to total income, which increased significantly for all types of banks except the CCBs. However, it should be noted that income from other services only represented around 10% of total income, suggesting that income predominantly was generated through lending activity.

To assess the relative importance of individual banks in Chinese loan markets, we have calculated the average market share of total loans (provided by all banks) per individual SOCB, JSCB, CCB and foreign bank. SOCBs had by far the largest individual share of total lending at around 20% of the full sample period, while those of individual CCBs and foreign banks were below 1%. The average market share of individual SOCBs dropped by nearly 5% in the post-WTO era compared with the pre-WTO era, which was the result of the increasing entry of new banks, as more CCBs were established and more foreign banks were allowed to start and expand business in China. Notwithstanding these new entries, individual JSCBs increased on average their market shares in loan markets post-WTO, reflecting their growing importance in lending activity. The average market share of total lending by individual CCBs and foreign banks decreased, which mainly reflects the fact that the number of these banks increased significantly in the post-WTO era.

The ratio of other non-earning assets to total assets decreased on average significantly for all types of banks, possibly reflecting improving profitability. The funding mix, defined as the ratio of customer deposits to total funding (excluding bank deposits), stayed relatively stable for all bank types except foreign banks. The equity to assets ratio generally decreased mildly. The other income to interest income ratio increased significantly for SOCBs and foreign banks, reflecting the increasing importance of non-traditional banking activities for these groups.

As in other empirical investigations of Chinese banking markets, the most troublesome data are those on wages. Ideally, the wage rate is the ratio of personnel expenses to the number of staff. However, many banks do not provide information on the number of their staff members, and for some banks personnel expenses data are missing as well. Therefore, we need to find an appropriate proxy for wages. Some researchers replace the missing number of employees by assuming that its growth rate is equal to that of total assets for a given bank (Fu and Heffernan, 2007; Altunbas et al., 2000; Rezvanian and

Mehdian, 2002; Vander Vennet, 2002). This approach might not be appropriate for our sample, as very few CCBs report the number of employees, so its growth rate cannot be calculated anyway. We instead follow the approach taken by Van Leuvensteijn et al. (2011) and proxy wages by the ratio of personnel expenses to total assets. We have complete data on total assets; so, to generate this proxy, we only need to have relatively complete data on personnel expenses. We adopted the following procedure to approximate missing data. For banks that provide these data but not for all years, we fill in the missing values of personnel expenses by assuming that they grew at the same rate as non-interest expenses. This is a reasonable assumption, as, according to Chinese accounting standards, non-interest expenses are composed of personnel expenses and non-operating expenses. For banks that do not report personnel expenses at all, we replace missing values by assuming that the ratio of personnel expenses to non-interest expenses equals the average of this ratio for the corresponding bank group, namely peit = peit/nieit * nieit, where peit/nieit is the average personnel expenses to non-interest expenses ratio, by bank type and year; j (j = SOCBs, JSCBs, CCBs, FOREIGN) represents bank groups and i stands for individual banks. Since our sample has almost complete data on non-interest expenses, we can use this approach to back-engineer the missing data on personnel expenses.

6 Results²²

6.1 Empirical results: Lerner and elasticity-adjusted Lerner indices

The estimation results for the cost equation (4.2) and supply equation (4.7) that form the basis for estimating the elasticity-adjusted Lerner index are reported in Appendix B, Table B.1. We summarise the results for the traditional Lerner index and elasticity-adjusted Lerner index in Table 4.

The results for both the traditional Lerner index and the elasticity-adjusted Lerner index suggest a general increasing level of bank competition up to around 2002 and a decreasing level of bank competition afterwards. Moreover, the traditional Lerner index indicates a lower level of competition than the elasticity-adjusted Lerner index for most years. Furthermore, the elasticity-adjusted Lerner index is significantly different from zero and one for all years, rejecting the null hypothesis that Chinese loan markets are in a state of either perfect competition or monopoly.

Turning to subsample estimates, the estimated values of the elasticity-adjusted Lerner index are significantly different from zero and one for each subsample as well. The results for both the traditional and elasticity-adjusted Lerner indices suggest that bank competition in the post-WTO period was lower than in the pre-WTO period, with the lowest level of competition registered for both indices in 2007. To test this conclusion formally, we conducted Chi-squared distributed Wald tests with one degree of freedom to determine whether the results of the elasticity-adjusted Lerner index are significantly different across the subsamples. Test statistics confirm that the elasticity-adjusted Lerner index for the pre-WTO period is lower than that for the post-WTO period, indicating strong evidence that competition worsened after WTO accession.

Lerner Inde	Table 4						
	λ_t	Average Ioan rate	Average Ioan deposit spread	Elasticity- adjusted Lerner index	Lerner index	Мс _е	MC
1997	0.062	0.168	0.085	0.367	0.358	0.095	0.098
1998	0.045	0.118	0.063	0.381	0.359	0.079	0.076
1999	0.029	0.102	0.044	0.288	0.305	0.064	0.061
2000	0.021	0.098	0.049	0.212	0.303	0.056	0.053
2001	0.019	0.086	0.050	0.224	0.303	0.052	0.049
2002	0.015	0.071	0.049	0.214	0.306	0.045	0.041
2003	0.017	0.071	0.049	0.235	0.327	0.047	0.043
2004	0.020	0.068	0.047	0.288	0.340	0.047	0.043
2005	0.023	0.079	0.054	0.287	0.358	0.052	0.048
2006	0.025	0.079	0.052	0.324	0.380	0.052	0.048
2007	0.036	0.079	0.052	0.452	0.434	0.050	0.046
2008	0.041	0.094	0.061	0.439	0.402	0.060	0.055
1996-2001	0.029	0.115	0.059	0.249	0.320	0.071	0.066
2002-2008	0.027	0.079	0.052	0.342	0.375	0.051	0.047

H 0: Elasticity A dj Lerner prewto> = Elasticity A dj Lerner postwto: chi²(1) = 12.13 p-value = 0.0002

 λ_t are statistically different from zero for all years at 1% significance level; *prewto* represents 1996-2001 period; *postwto* represents 2002-2008 period;

MCe and MCI are average marginal costs used to calculate elasticity-adjusted Lerner index and traditional Lerner index, respectively.

22.

The results for the Panzar-Rosse H-statistic are reported in Appendix A.

Our results are reinforced by Soedarmono et al. (2013) and Fungáčová et al. (2012),²³ which also document a general decreasing trend of bank competition in China during 2002-2008. The latter study obtains an average Lerner index of 0.378 for this period, while our elasticity-adjusted Lerner index and traditional Lerner indices for the same period are 0.342 and 0.375, respectively. Comparison with results from studies for other countries show that the values obtained for China are relatively high: Berger et al. (2009b) found an average Lerner index of 0.22 for 23 industrial countries calculated over the period 1999-2005, while Carbó Valverde et al. (2009) obtained a mean of 0.16 for the European Union during 1995-2001. At the same time, our estimates for the pre-WTO period are around 0.25, indicating that competition in Chinese loan markets was not that much lower than that in developed economies during those years. The post-WTO period, however, is significantly less competitive for China than for the other countries.

Therefore, regardless of whether the elasticity-adjusted Lerner index or traditional Lerner index is adopted as the indicator to measure competition, we conclude that Chinese loan markets were relatively competitive in the pre-WTO period, while competitive conditions worsened later.

6.2 Empirical results: Price Elasticity (PE) indicator

Similarly to the Lerner index, the empirical estimation of the PE indicator starts with the estimation of marginal costs. In this section, we improve the marginal cost estimation by assuming different Translog Cost Functions (TCF) for each bank type. More specifically, we estimate one separate TCF for the SOCBs, JSCBs, CCBs and the foreign banks, which should improve the accuracy of the estimation of marginal costs. The estimation of the TCF is reported in Appendix C, Table C.1. Given the estimated marginal costs, we are now able to estimate the PE indicator. For China, we use the relationship between the marginal costs of individual banks and their profits:

$$\ln \pi_{ilt} = \alpha + \sum_{t=1,\dots,T} \gamma_t d_t + \sum_{t=1,\dots,T} \beta_t d_t \ln mc_{ilt} + u_{ilt}$$
(6.1)

where π_{ilt} stands for profits, d_t is a time dummy, mc_{ilt} denotes marginal costs, *i* refers to bank *i*, *l* to output type "loans", and *t* to year *t*; u_{ilt} is the error term. This provides us with the coefficient β_t , ie the PE indicator (as is given by PE = $d(\ln \pi_t)/d(\ln mc_t)$). β_t is negative in theory, reflecting that higher marginal costs reduce profits for all banks.²⁴ Moreover, the more competitive the market is, the lower the value of β_t . In other words, banks are punished more harshly for being inefficient in more competitive markets. Note that the indicator β_t is timedependent.

Profits are defined as:

$$\pi_{ilt} = \mathbf{x}_{ilt} \left(p_{ilt} - mc_{ilt} \right) \tag{6.2}$$

where x_{ilt} denotes the total amount of loans and p_{ilt} is the loan interest rate calculated as interest income over loans.

^{23.} Soedarmono et al. (2013) used the elasticity-adjusted Lerner index, while Fungáčová et al. (2012) estimated the traditional Lerner index.

^{24.} In practice, a positive *t* is possible (Van Leuvensteijn et al., 2007), which could be the result of extreme collusion, market regulation or banks competing on quality (Tabak et al., 2012).

We expect higher profits to go hand in hand with lower marginal costs, but since our definition of profits is a function of marginal costs, there may be an endogeneity problem. To correct for this, we employ lagged marginal costs as instrument variable and investigate various alternative estimation techniques.

We follow the strategy set out by Angrist and Pischke (2009) and first test whether the instrumental variables are weak. For this purpose, we employ Angrist-Pischke (AP) Fstatistics to test for weak identification of individual endogenous regressors. The AP firststage F-statistics indicate that a particular endogenous regressor is weakly identified if the null hypothesis is rejected.²⁵ Table 5 reports that nearly all instrumental variables used are strong with F-test values above 16.38, with the exception of the instrumental variables for 1997, 1998, 2002 and 2003, indicating for these years weak instrumental variables.

Because the instrumental variables for some years have weak properties, we use only just-identified instruments as they are median-unbiased and not subject to the weak instrumental variable critique. Furthermore, following the suggestion of Angrist and Pischke (2009), we check the two-stage least squares (2SLS) results with estimates from Limited Information Maximum Likelihood (LIML), as the latter results are less biased. LIML can be seen as a "combinatory estimation" technique where the ordinary least square (OLS) and 2SLS estimations are combined and the weights for the two estimations are determined by the data (see Angrist and Pischke, 2009, for further explanation). We use as instrument variables one-year lagged values of marginal costs and kernel-based heteroskedastic and autocorrelation-consistent (HAC) variance estimations. The bandwidth in the estimation is set at two periods, and the Newey-West kernel is applied. The results of 2SLS and LIML are very similar, in fact almost identical, and therefore we only present the results with LIML.²⁶

To assess the evolution of bank competition, we first estimate the yearly PE indicator, which is based on equation (6.1). Table 5 reports the results. The yearly PE indicators are significantly different from zero for most of the sample years, except for the 1997-2000 period. Competition increased sharply during 2001-2003 and then declined up to 2005. It then intensified again, followed by a slightly decreasing level of competition in 2007 and 2008. In general, the development of the yearly PE indicator suggests that competitive conditions in Chinese Ioan markets improved, especially after WTO accession in 2001. Admittedly, the insignificant results for the early years in our sample could be caused by the small number of observations for those years, in which case the results could be influenced strongly by outliers. Therefore, we estimate the PE indicator for subsamples to avoid small-sample bias.

^{25.} The Stock-Yogo weak ID test critical value at 10% (maximal LIML size) is 16.38 (Stock and Yogo, 2005).26. Results with 2SLS are available upon request.

Yearly	PE Indicator	
--------	--------------	--

Dependent variable: In(Profits) Table									
	PE Indicator	z-value	AP chi ² (1) p-value	AP F (1,433)					
1997	5.783	(0.44)	0.4866	0.46					
1998	-2.177	(-1.23)	0.1021	2.53					
1999	1.489	(0.56)	0.0000	31.78					
2000	0.147	(0.05)	0.0000	27.91					
2001	-4.250***	(-5.85)	0.0000	31.11					
2002	-5.497**	(-2.36)	0.0002	13.10					
2003	-6.327***	(-2.64)	0.0147	5.64					
2004	-4.092***	(-3.92)	0.0000	58.28					
2005	-1.352	(-1.45)	0.0000	67.26					
2006	-4.024***	(-4.17)	0.0000	20.73					
2007	-3.611***	(-5.36)	0.0000	89.77					
2008	-2.482***	(-4.12)	0.0000	28.15					
Constant	0.401	(0.23)							
Nr obs		4	57						
F		6.2	49						
Centered R ²		0.1	31						

z-values in parenthesis; ** represents significance level of 5%, *** represent significance level of 1%. AP chi² is the Angrist-Pischke (AP) first-stage chi-squared test. AP F is the Angrist-Pischke (AP) F-statistic, which can be compared to Stock-Yogo (2002) and (2005) critical values for Cragg-Donald F statistic with K1=1. The Stock-Yogo weak ID test critical value at 10% maximal LIML size is 16.38. Year dummies are not reported to save space.

The subsamples are defined in the same way as in the previous sections, eg pre-WTO (1996-2001) and post-WTO (2002-2008). We estimate one PE indicator for each subsample and test whether competition changed significantly after WTO accession. These point estimates can be interpreted as averages of yearly estimates over their respective sample periods, weighted by the number of observations in each year. A point estimate of the PE indicator for the whole sample 1996-2008 is provided as well. Estimations are based on the following equation:

 $\ln \pi_{ilt} = \alpha + \gamma Trend + \beta \ln mc_{ilt} + u_{ilt}$

(6.3)

where Trend is a time trend.27

Table 6 reports the results for the subsample estimations. The Kleibergen-Paap rk LM statistics are significant at the 1% level for the whole sample and each subsample, rejecting the null hypothesis that the model is unidentified. The Kleibergen-Paap Wald rk Fstatistic for each sample is larger than 16.38, suggesting that the estimations do not suffer from weak identification. Both test statistics are robust to heteroskedasticity and autocorrelation. All PE indicators have the correct sign (negative) and are significant at the 1% level, except for the pre-WTO period. To test whether competitive conditions in Chinese loan markets experienced significant structural change after WTO accession, we performed a Chisquared distributed Wald test with one degree of freedom to determine whether the PE indicators are significantly different across various subsamples. The rejection of the null hypothesis (H0: pre-WTO PE indicator <= post-WTO PE indicator) indicates that the level of bank competition was significantly higher in the post-WTO period.

^{27.} Using year dummies instead of a time trend generates similar results for all estimations reported in this paper. Results are available upon request.

Point estimates PE indicator: Whole sample and subsamples

Dependent variable: In(Profits)			Table 6
	1996-2008	1996–2001	2002–2008
PE Indicator	-2.388***	-1.514	-3.570***
	(-5.78)	(-1.43)	(-7.74)
Time Trend	-0.0332	-0.519**	0.345***
	(-0.82)	(-2.37)	-4.71
Constant	-0.24	4.966**	-8.050***
	(-0.19)	(2.18)	(-4.51)
H ₀ :prewto -postwto <=0(p-value)		3.61** (0.0288)	
Nr. Obs	457	87	370
F	16.78	2.97	33.67
Centered R ²	0.089	0.141	0.18
Kleibergen-Paap rk Wald F	211.4	30.98	130.8
Kleibergen-Paap rk LM (p-value)	62.00(0.0000)	13.94(0.0001)	44.22(0.0000)

Dependent variable: In(Profits)

z-values in parenthesis; ** represents significance level of 5%, *** represents significance level of 1%. Since there is only one endogenous variable, we use Kleibergen-Paap rk Wald F and Kleibergen-Paap rk LM tests to test weak identification and under-identification. The Stock-Yogo weak ID test critical value at 10% maximal LIML size is 16.38.

Finally, our estimates of the yearly PE indicators and of the PE indicators for the whole samples and subsamples (point estimates) are robust to different estimation methods and different specifications of the PE indicator. These robustness tests are presented in Appendix E, for an alternative definition of the PE indicator (E.4) and for an alternative calculation of marginal costs (E.5). The latter analysis shows that the slightly different calculation of marginal costs for the (elasticity-adjusted) Lerner index and the PE indicator does not drive the divergence between their results.

6.3 Comparison of the various empirical measures

The results for the various measures that we presented in the previous sections revealed significant differences in the evolution of competition and its level, for both the yearly and subsample estimates. The fact that different competition measures yield inconsistent results for the same banking market and country is well documented in the empirical banking literature, as discussed in Sections 1 and 2. At the same time, this finding may be especially relevant for relatively regulated markets such as Chinese loan markets. Hence, following Carbó Valverde et al. (2009), we formally test for the consistence of different competition measures by calculating pair-wise correlation coefficients. Since all competition measures except the H-statistic imply higher competition with a lower value, we multiply the H-statistic results by -1 to make comparison easier, so that now a higher value implies lower competition for all measures. The results of the pair-wise correlations are reported in Table 7.

Pair-wise correlation coefficients Table											
	PE	Н	Lerner	Elasticity-adjusted Lerner							
PE	1										
Н	-0.1884*	1									
Lerner	-0.0413	0.4154*	1								
Elasticity-adjusted											
Lerner	0.1433*	0.3522*	0.9321*	1							
TIME	-0.4152*	0.2779*	0.7602*	0.5869*							
* represents significance le	vel of 1%.										

Testing the pair-wise correlations of these measures of competition first reveals that the PE indicator is negatively correlated with the H-statistic and the traditional Lerner index (the latter not significant at 1%). This finding confirms that the former indicator yields diametrically opposed results to those from the latter two traditional measures. At the same time, the PE indicator is positively correlated with the elasticity-adjusted Lerner index. Second, in order to test whether the competition measures produce similar conclusions on the evolution of competition over time, we provide pair-wise correlation coefficients with time. A negative (positive) value indicates improved (worsened) competition over time. The PE indicator suggests improving competition across the sample years, while the other measures suggest the opposite, a result that is consistent with our results in the previous sections.

6.4 Interpretation results PE indicator

As we have argued above, the PE indicator is the most appropriate measure to assess competitive conditions in Chinese loan markets. This section provides an in-depth analysis of its results. Overall, we are generally well able to explain the specific development of this indicator over time, which strengthens our belief that it is superior to the Panzar-Rosse and Lerner approaches in the context of China. In fact, we find it rather difficult to offer plausible explanations for the results obtained with the latter two measures.

The key to understanding why the results for the PE indicator are so different from those of more conventional competition indicators like the Panzar-Rosse H-statistic and the (elasticity-adjusted) Lerner index lies in the system of interest rate regulation in China. If interest regulation is binding, it can substantially bias the traditional measures but not RPD. We have proved this theoretically for the Lerner index in Section 4.3. In the case of the H-statistic, we discuss this bias in Appendix A (A.3).

Whether and to what extent interest regulation is binding is an empirical question which definitely requires more attention in the literature on measuring bank competition. The empirical literature on binding interest rate regulation in the context of China is rather small. However, the general consensus is that: a) the lending rate floor is considered to be non-binding in practice (He and Wang, 2012; see Section 3.1); b) the deposit rate floor and ceiling are binding (Feyzioğlu, 2009; Ma et al., 2011; He and Wang, 2012;²⁸ PBC, 2009; Yi, 2009); and c) the lending rate ceiling was most likely binding during the pre-WTO period (Yi, 2009). In this section, we argue that the bias in the H-statistic and Lerner index predominantly results

^{28.} He and Wang (2012, p. 34): "Using the regression results, we can then estimate the equilibrium interest rate by subtracting the effects of financial repression from the observed real interest rate: the equilibrium deposit rate in China was estimated at 4.7% in 2005. This estimated equilibrium deposit rate is significantly higher than the observed real deposit rate of 1.6% in 2005, which means that the deposit-rate ceiling must have been binding in China."

from binding interest rate regulation. The interest rate regulation discussed here is mainly the ceilings on deposit and lending rates, but similar analysis can be extended to the floors applied to these rates.

Turning to the specific results we obtained in the PE indicator estimations, we find generally positive and insignificant values for the PE indicator for the early years of our sample, suggesting that during 1997-2000 a negative relationship between efficiency (marginal costs) and profits could not be established (see Section 6.2, Table 5). We are encouraged by this result, as one would expect that, during the years when Chinese banking markets were still heavily regulated, more efficient banks would not be more profitable, with competition being suppressed. In other words, there was no reward for being more competitive than one's competitors. Actually, this finding reminds us of the results for Japanese loan markets during the 1990s in Van Leuvensteijn et al. (2007 and 2011), when the PE indicator was positive (and significant). This could be related to the regulated "convoy system" in Japan where market shares were more or less guaranteed and competitive forces were largely absent.

Subsequently, we start to find negative and highly significant values for the PE indicator for Chinese loan markets from 2001 onwards, indicating that, as loan markets became more competitive, more efficient banks started to generate more profits than less efficient banks. The PE indicator improved further until 2003, when it reached its lowest value of –6.3 (eg highest level of competition). From an international perspective, this value is comparable to the most competitive yearly results we obtained for several mature economies (Van Leuvensteijn et al., 2007 and 2011).

Then, after 2003, we find a gradual decline in competition in Chinese loan markets (but still with negative and, except for one year, highly significant results), which was the most notable in 2004, 2007 and 2008. We believe that various policy measures and a certain degree of re-regulation may be responsible for this pattern of slightly declining competition. In general, there is evidence that, for both mature and emerging market economies, financial deregulation has often been intertwined with concomitant prudential re-regulation (Zhao et al., 2010). This seems also to have occurred to a certain extent in China. In 2004, the CBRC adopted new capital adequacy requirements, including the requirement to fully provision their non-performing loans and maintain at least 8% of aggregate capital adequacy, that banks should meet by 2007 (Podpiera, 2006). Further in 2004, the CBRC strengthened other parts of its regulatory policies, including its on-site examinations and monitoring of large exposures, and introduced risk-based supervision for the CCBs. Regulation was tightened regarding non-performing loans (NPLs), with a view to reducing banks' NPL ratios (Liu, 2009). The combined impact of these measures may have affected competitive conditions in Chinese loan markets. In addition, the PBC, worried by a possible overheating of the Chinese economy, re-introduced credit quotas in the fall of 2007 that aimed to mitigate bank lending growth. As formulated by Fukumoto et al. (2010, p. 3): "The newly introduced credit limits were similar to the credit plan which had existed until 1998 in the sense that both of these measures set rigorous constraints on the growth of bank lending. The growth of bank lending started to slow down once credit limits were implemented." These lending restrictions were kept in place until the fall of 2008 and can be characterised as a major step of re-regulation, as they re-instated elements of the old credit plan system. It may be regarded that this policy move had a detrimental impact on bank competition in China, as it frustrated commercially oriented lending decisions and disincentivised competition.

The element of re-regulation is picked up nicely by the financial repression index developed for China in Huang and Wang (2011) (see Figure 3, left-hand panel). It is based on six financial repression variables, including two interest rates, two loan market share variables, reserve requirements and capital account controls. During the years of our sample – 1996-2008 – the index declines, suggesting less financial repression for all years except for 2004, 2007 and 2008, when it increases. After its first rise in 2004, indicating stronger financial repression, it fell to its lowest level ever in 2006, before strongly increasing in 2007, followed by a further pick-up in 2008. The yearly results of the PE indicator, which are depicted for illustrative purposes in Figure 3 (left-hand panel), closely follow the pattern of the financial repression index. The generally increasing re-regulation in the latter part of our sample may be reflected in the rather sharply increasing deposits to loans ratio from 2004 onwards (Figure 3, right-hand panel). Possibly, tightened loan controls and other regulatory steps forced banks to reduce the growth of their loans relative to that of their deposits.



Given the strong similarity between the pattern of the PE indicator and the financial repression index, we are interested in how this relation looks for other financial liberalisation indices. To this end, we employ two additional indicators of financial reform: the overall financial liberalisation and interest rate liberalisation indices developed by Abiad et al. (2010). Their values are shown in Appendix D, Table D.1. The former index measures the overall degree of financial liberalisation, with values ranging from 0 to 1, with a higher value indicating a more liberalised financial system. The latter index, which takes the values 0, 1, 2 or 3, indicates fully repressed, partially repressed, partially liberalised and fully liberalised interest rates, respectively.

In order to provide a more comprehensive analysis, we calculate the pair-wise correlation coefficients between the three indices of financial reform and the same four measures of competition that we used in Section 6.3.

Again, we multiply the H-statistic results by -1 to make comparison of the correlations easier. The results are reported in Table 8. Should financial reform promote competition, one would expect positive correlations between the financial repression index and the competition measures. This is because both the index and the measures show improved conditions with lower values and vice versa. In contrast, one would expect negative correlations between the two other financial liberalisation indices and the competition

measures if financial reform promotes competition. Namely, the two indices indicate a more liberalised financial system with higher values, while the indicators of competition suggest more competition with lower values. Since financial reform may affect banking behaviour with a time lag, we use one-period lagged values of the three indices.²⁹

Pair-wise correlation coefficients with financial reform indices Table 8												
	PE	н	Lerner	Elasticity-adjusted Lerner								
FREP	0.6560*	-0.2159*	-0.5794*	-0.3104*								
Fin_Lib Index	-0.4223*	0.2689*	0.5015*	-0.055								
Int_Lib Index	-0.2206*	0.4175*	0.7917*	0.3285*								

* represents significance level of 1%. FREP is the financial repression index as reported in Huang and Wang (2011). Fin_Lib Index and Int_Lib_Index represent financial liberalization index and interest rate liberalization index, respectively. The values of the indices are reported in Appendix D, Table D.1.

In case a more liberalised financial system is associated with more intense competition, the correlations of the PE indicator show the expected sign with all three indices (positive for the financial repression index and negative for the two others). The correlations are also highly significant at the 1% level. In contrast, the correlations of the other measures (ie the H-statistic and the Lerner indices) that are significant all have the opposite sign, suggesting that increased liberalisation is associated with weaker competition.

Although we did not formally test the impact of financial reform on competition in Chinese loan markets, the pair-wise correlation coefficients that we find for the PE indicator, all associating more reform with more competition, tend to bolster our confidence in it. The empirical literature on the relationship between financial reform and competition is not always conclusive, but generally its results for emerging market economies have shown a beneficial link between the two (see Section 2). That we find this for China is in our view encouraging.

Finally, a number of structural developments in the Chinese financial system should support increasing competition in Chinese banking. First, the market share of the four SOCBs in bank lending has declined steadily over the past decade. This should have contributed to improving competition. Second, China has experienced a steady rise of the shadow banking system, which should have increased competition in bank loan markets, as banks have to compete for a smaller share of total credit intermediation in China. Third, Chinese banks have introduced personal incentive-driven remuneration packages, which link the remuneration of individual loan officers much more to their actual performance. This should have contributed to competition in Chinese loan markets as well.

^{29.} We also employed the current values of the financial reform indices to account for the possibility that banks may anticipate financial reform measures and adjust their competitive strategies accordingly. The results are similar to the ones we report here.

7 Conclusions

This paper investigates the evolution of competition in Chinese Ioan markets. We believe that this investigation makes sense, as after 30 years of financial reform, China's banks have begun to behave more like commercial banks in mature economies (Firth et al., 2009; Herd et al., 2010; IMF, 2011). The impact of financial reform on bank competition has been investigated extensively for many countries, and for most mature and emerging economies empirical studies suggest that it promoted bank competition.

However, some earlier studies that adopted conventional approaches to measure competition concluded that bank competition in China declined during the past decade, despite these reforms. In this paper, we compare the results obtained from conventional indicators such as the Lerner index and Panzar-Rosse H-statistic with those estimated using the relatively new Profit Elasticity (PE) approach. We argue that traditional measures of competition fail to measure competition in the Chinese banking sector properly, and we provide arguments – both theoretically and empirically – to support this.

Using balance sheet information for a large sample of banks operating in China during 1996-2008, we show that competition actually increased in the past decade when the PE measure introduced by Boone et al. (2007) and Boone (2008) is used as indicator of competition. We find that the period after China's entry into the WTO in 2001 was characterised by significantly more competitive loan markets than before. This stands in contrast to the results that we obtain by calculating the conventional and elasticity-adjusted Lerner indices and the H-statistic. We doubt these findings, as they may be distorted by various factors, including restrictions on market shares and interest rates.

This study yields two major insights. First, the theoretical foundation of the PE indicator, which is the RPD model, is not biased due to interest rate regulation. This makes the PE indicator a much better measure to gauge competition in Chinese loan markets than conventional approaches. This is a very general insight that can be useful for investigations of competitive conditions in banking markets in countries where binding regulation of interest rates is a distinctive characteristic. Second, applying this unbiased competition indicator to Chinese loan markets shows that financial reform indeed has contributed to significant improvements in competition. This result is much in line with those obtained for other emerging economies. Again, we find contradictory results for the conventional measures. Moreover, our results for both the PE indicator and the other measures are robust for a large number of alternative specifications and estimation methods.

All in all, our analysis suggests that bank lending markets in China have been more competitive than previously assumed. It may also provide an answer to questions raised in other research on this issue. For example, Fungáčová et al. (2012) use the Lerner index and find results similar to ours for the same index, indicating that bank competition in China declined over time. They go on to note that "at first glance, it is somewhat remarkable that China's accession to WTO has not led to greater competition in the banking industry". As we have argued in this paper, this may not be that remarkable after all. It may be the case that competition in the Chinese banking system has not been assessed with the proper method. Of course, further work is definitely needed to substantiate this claim. For example, it would be interesting to see how the results that we report here compare with those for other

banking systems of comparable structure and stage of development to China's. Moreover, given the strong differences in regional economic and financial development across China, the PE indicator and conventional measures could be estimated for the regional banking market in China. For those that de facto operate under more liberalised conditions, the differences between the PE indicator and other measures may be much smaller, for example when compared with those for much less developed and de facto more regulated regions.

More generally, our findings for China indicate that the bank competition literature may wish to focus more explicitly on the potential biases in competition measures that result from the existence of interest rate and other regulations. Regarding the former, empirical work to assess whether, and to what extent, interest rate regulation is binding is crucial to validating whether conventional measures of competition may obtain unbiased results.

REFERENCES

ABIAD, A., E. DETRAGIACHE AND T. TRESSEL (2010), "A new database of financial reforms", *IMF Staff Papers*, 57(2), 281-302.

ALLEN, F., J. QIAN AND M. QIAN (2005), "Law, finance, and economic growth in China", *Journal of Financial Economics*, 77(1), 57-116.

ALLEN, F., J. QIAN, C. ZHANG AND M. ZHAO (2012), "China's financial system: Opportunities and Challenges", NBER Working Paper No.17828.

ALTUNBAS, Y., M. LIU, P. MOLYNEUX AND R. SETH (2000), "Efficiency and risk in Japanese banking", *Journal of Banking and Finance*, 24, 1605-1628.

ANGELINI, P. AND N. CETORELLI (2003), "The effects of regulatory reform on competition in the banking industry", Journal of Money, Credit and Banking, 35(5), 663-684.

ANGRIST, J. AND J.-S. PISCHKE (2009), *Mostly harmless econometrics*, Princeton and Oxford: Princeton University Press.

APPELBAUM, E. (1982), "The estimation of the degree of oligopoly", Journal of Econometrics, 19, 287-299.

- ARIFF, M. AND L. CAN (2008), "Cost and profit efficiency of Chinese banks: A non-parametric analysis", China Economic Review, 19, 260-273.
- BARROS, C., Z. CHEN, Q. LIANG AND N. PEYPOCH (2011), "Technical efficiency in the Chinese banking sector", *Economic Modelling*, 28(5), 2083-2089.

BEATTIE, B. AND C. TAYLOR (1985), The Economics of Production, New York: John Wiley & Sons.

BERGER, A., I. HASAN AND M. ZHOU (2009a), "Bank ownership and efficiency in China: What will happen in the world's largest nation?", *Journal of Banking and Finance*, 33(1), 113-130.

- (2010), "The effects of focus versus diversification on bank performance: Evidence from Chinese banks", Journal of Banking and Finance, 34(7), 1417-1435.
- BERGER, A., L. KLAPPER AND R. TURK-ARISS (2009b), "Bank competition and financial stability", Journal of Financial Services Research, 35(2), 99-118.

BERGER, A. AND L. MESTER (1997), "Inside the black box: What explains differences in the efficiencies of financial institutions?", *Journal of Banking and Finance*, 21(7), 895–947.

BIKKER, J. (2004), Competition and efficiency in a unified European banking market, Cheltenham: Edward Elgar.

 (2010), "Measuring performance of banks: An assessment", Journal of Applied Business and Economics, 11(4), 141-159.

BIKKER, J. AND K. HAAF (2002), "Competition, concentration and their relationship: An empirical analysis of the banking industry", *Journal of Banking and Finance*, 26, 2191-2214.

BIKKER, J. AND J. BOS (2005), "Trends in competition and profitability in the banking industry: A basic framework", SUERF Studies, 2005/2.

BIKKER, J., L. SPIERDIJK AND P. FINNIE (2007), "Misspecification of the Panzar-Rosse Model: Assessing competition in the banking industry", unpublished mimeograph, 30 July.

- BIKKER, J. AND L. SPIERDIJK (2008), "How banking competition changed over time", De Nederlandsche Bank Working Paper No.167.
- BIKKER, J. AND M. VAN LEUVENSTEIJN (2008), "Competition and efficiency in the Dutch life insurance industry", *Applied Economics*, 40(16), 2063-2084.

BIKKER, J., S. SHAFFER AND L. SPIERDIJK (2012), "Assessing competition with the Panzar-Rosse Model: The role of scale, costs and equilibrium," *Review of Economics and Statistics*, 94(4), 1025-1044.

BOONE, J. (2000), "Competition", CEPR Discussion Paper Series No. 2636.

 (2001), "Intensity of competition and the incentive to innovate", International Journal of Industrial Organization, 19(5), 705–726.

- (2008), "A new way to measure competition", Economic Journal, 118, 1245 -1261.

BOONE J., R. GRIFFITH AND R. HARRISON (2004), "Measuring competition", paper presented at the Encore Meeting 2004 "Measuring competition".

BOONE, J., J. VAN OURS AND H. VAN DER WIEL (2007), "How (not) to measure competition", Tilburg University, Tilburg Law and Economics Center (TILEC) Discussion Paper No.2007-014.

BOONE, J. AND M. VAN LEUVENSTEIJN (2010), "Measuring competition using the profit elasticity: American sugar industry, 1890-1914", CEPR Discussion Paper Series No.8159.

BRANDT, L. AND T.G. RAWSKI (2008), "Chinese economic reforms", in: S. Durlauf and L. Blume (eds.), *The New Palgrave Dictionary of Economics*, second edition, London: Palgrave Macmillan.

BRESNAHAN, T. (1982), "The oligopoly solution concept is identified", Economics Letters, 10, 87–92.

CARBÓ VALVERDE, S., D. HUMPHREY, J. MAUDOS, AND P. MOLYNEUX (2009), "Cross-country comparisons of competition and pricing power in European banking", *Journal of International Money and Finance*, 28(1), 115-134.

CASU, B. AND C. GIRARDONE (2006), "Bank competition, concentration and efficiency in the Single European Market", The Manchester School, 74(4), 441-468.

- (2009), "Competition issues in European banking," Journal of Financial Regulation and Compliance, 17(2), 119-133.

CENTRAAL PLANBUREAU (CPB) (2000), "Measuring competition", paper presented at *The ENCORE Meeting 2004* "Measuring Competition".

CHANG, T.-P., J.-L. HU, R. CHOU AND L. SUN (2012), "The sources of bank productivity growth in China during 2002-2009: A disaggregation view", *Journal of Banking and Finance*, 36(7), 1997-2006.

CHEN, Q., M. FUNKE AND M. PAETZ (2012), "Market and non-market monetary policy tools in a calibrated DSGE model for mainland China", Bank of Finland, BOFIT Discussion Papers No.16/2012.

CHENG, X. AND H. DEGRYSE (2010), "The impact of bank and non-bank financial institutions on local economic growth in China", *Journal of Financial Services Research*, 37, 179-199.

CLAESSENS, S. (2009), "Competition in the financial sector: Overview of competition policies", The World Bank Research Observer, 24(1), 83-118.

CLAESSENS, S. AND L. LAEVEN (2004), "What drives bank competition? Some international evidence", *Journal of Money, Credit and Banking*, 36(3), Part 2, 563-583.

CORTS, K. (1999), "Conduct parameters and the measurement of market power", Journal of Econometrics, 88, 227-250.

CULL, R. AND L. XU (2000), "Bureaucrats, state banks and the efficiency of credit allocation: The experience of Chinese state-owned enterprises", *Journal of Comparative Economics*, 28, 1-31.

DELATTE, A.-L. (2007), "Monetary policy and macrocontrol in China: The actual impact of window guidance", Hong Kong University of Science and Technology, mimeograph.

DELIS, M. (2012), "Bank competition, financial reform and institutions: The importance of being developed", *Journal of Development Economics*, 97, 450-465.

DEMSETZ, H. (1973), "Industry structure, market rivalry, and public policy", Journal of Law and Economics, (16),1-9.

DOBSON, W. AND A. KASHYAP (2006), "The contradiction in China's gradualist banking reforms", *Brookings Papers on Economic Activity*, 37(2), 103-162.

DU, Y. (2010), "Monetary policy and bank loan supply in China", thesis Williams College, 14 May.

FERNÁNDEZ DE GUEVARA, J., J. MAUDOS AND F. PÉREZ (2005), "Market power in European banking sectors", Journal of Financial Services Research, 27(2), 109–137.

 (2007), "Integration and competition in European financial markets", Journal of International Money and Finance, 26(1), 26–45.

FERRI, G. (2009), "Are New Tigers supplanting Old Mammoths in China's banking system? Evidence from a sample of city commercial banks", *Journal of Banking and Finance*, 33(1), 131-140.

FEYZIOĞLU, T. (2009), "Does financial performance mean good financial intermediation in China?", IMF Working Paper No.WP/09/170.

FEYZIOĞLU, T., N. PORTER AND E. TAKÁTS (2009), "Interest rate liberalisation in China", IMF Working Paper No.WP/09/171.

- FIRTH, M., C. LIN, P. LIU AND S. WONG (2009), "Inside the Black Box: Bank credit allocation in China's private sector", Journal of Banking and Finance, 33(6), 1144-1155.
- FU, X. AND S. HEFFERNAN (2007), "Cost X-efficiency in China's banking sector", China Economic Review, 18, 35-53.
- (2009), "The effects of reform on China's bank structure and performance", *Journal of Banking and Finance*, 33(1), 39-52.

FU, X. (2009), "Competition in Chinese commercial banking", Banking and Finance Review, 1(1), 1-16.

- FUKUMOTO, T., M. HIGASHI, Y. INAMURA AND T. KIMURA (2010), "Effectiveness of window guidance and financial environment: In light of Japan's experience of financial liberalisation and a bubble economy", Bank of Japan, *Review*, No.2010-E-4.
- FUNGÁČOVÁ, Z., P. PESSAROSSI AND L. WEILL (2012), "Is bank competition detrimental to efficiency? Evidence from China", Bank of Finland, BOFIT Discussion Papers No.31/2012.
- GARCÍA-HERRERO, A., S. GAVILÁ AND D. SANTABÁRBARA (2005), "China's banking reform: An assessment of its evolution and possible impact", Banco de España Documentos Ocasionales No.0502.
- (2009), "What explains the low profitability of Chinese banks?", Journal of Banking and Finance, 33, 2080-2092.

GARCÍA-HERRERO, A. AND D. SANTABÁRBARA (2008), "Does the Chinese banking system benefit from foreign investors?", Bank of Finland, BOFIT Discussion Papers No.11/2008.

GEIGER, M. (2008), "Instruments of monetary policy in China and their effectiveness: 1994-2006", UNCTAD Discussion Papers No.187.

GENESOVE, D. AND W. MULLIN (1998), "Testing static oligopoly models: Conduct and cost in the sugar industry, 1890-1914", *The RAND Journal of Economics*, 29(2), 355-377.

GOODFRIEND, M. AND E. PRASAD (2006), "A framework for independent monetary policy in China", IMF Working Paper No.WP/06/111.

GREEN, S. (2005), "Making monetary policy work in China: A report from the money market front line", Stanford University, Stanford Center for International Development Working Paper No.245.

GUTIÉRREZ DE ROZAS, L. (2007), "Testing for competition in the Spanish banking industry: The Panzar-Rosse approach revisited", Banco de España, Documentos de Trabajo No.0726.

- HASAN, I. AND R. XIE (2012), "A note on foreign bank entry and bank corporate governance in China", Bank of Finland, BOFIT Discussion Papers No.8/2012.
- HE, L. AND X. FAN (2004), "Foreign banks in post-WTO China: An intermediate assessment", China and World Economy, 12(5), 3-16.
- HE, D. AND H. WANG (2011), "Dual-track interest rates and the conduct of monetary policy in China", Hong Kong Institute for Monetary Research Working Paper No.21/2011. Shortened version published as: He, D. and H. Wang (2012), "Dual-track interest rates and the conduct of monetary policy in China", *China Economic Review*, 23(4), 928-947.
- HEFFERNAN, S. AND X. FU (2010), "Determinants of financial performance in Chinese banking", *Applied Financial Economics*, 20(20), 1585-1600.
- HERD, R., S. HILL AND C. PIGOTT (2010), "China's financial sector reforms", OECD Economics Department Working Papers No.747.
- HONG, Z. AND Y. YAN (1997), "Trust and investment corporations in China", Federal Reserve Bank of Cleveland Working Paper No.97/06.
- HUANG, Y. (2010), "Dissecting the China puzzle: asymmetric liberalisation and cost distortion", Asia Economic Policy Review, 5, 281–295.

- HUANG, Y., X. WANG, B. WANG AND N. LIN (2010), "Financial reform in China: Progresses and challenges", mimeograph, Peking University, China Macroeconomic Research Center, 29 November.
- HUANG, Y. AND X. WANG (2011), "Does financial repression inhibit or facilitate economic growth? A case study of Chinese reform", *Oxford Bulletin of Economics and Statistics*, 73(6), 833-855.
- IMF (2011), People's Republic of China: Financial System Stability Assessment, November.
- JORGENSON, D. (1986), "Econometric methods for modelling producer behaviour", in: E. Grilliches and M. Intriligator (eds.), *Handbook for Econometrics, Vol.III*, Amsterdam: Elsevier Science Publishers, 1841-1915.
- KPMG (2011), Mainland China Banking Survey 2011, KPMG.
- KOETTER, M., J. KOLARI AND L. SPIERDIJK (2012), "Enjoying the quiet life under deregulation? Evidence from adjusted Lerner indices for U.S. banks", *The Review of Economics and Statistics*, 94(2), 462-280.
- KUMBHAKAR, S. AND D. WANG (2007), "Economic reforms, efficiency and productivity in Chinese banking", Journal of Regulatory Economics, 32(2), 105-129.
- LAU, L. (1982), "On identifying the degree of competitiveness from industry price and output data", *Economics Letters*, 10, 93–99.
- LARDY, N. (2006), "Comments and discussion", Brookings Papers on Economic Activity, 37(2), 149-152.
- LEUNG, M. AND R. CHAN (2006), "Are foreign banks sure winners in post-WTO China?", Business Horizons, 49, 221-234.
- LEUVENSTEIJN, M. VAN, J. BIKKER, A. VAN RIXTEL AND C. KOK-SØRENSEN (2011), "A new approach to measuring competition in the loan markets of the euro area", *Applied Economics*, 43(23), 3155-3167. Earlier published as: Van Leuvensteijn, M., J. Bikker, A. van Rixtel and C. Kok-Sørensen (2007), "A new approach to measure competition in the loan markets of the euro area", ECB Working Paper No.768.
- LEUVENSTEIJN, M. VAN, C. KOK-SØRENSEN, J. BIKKER AND A. VAN RIXTEL (2013), "Impact of bank competition on the interest rate pass-through in the euro area", *Applied Economics*, 7(48), 1359-1380. Earlier published as: Van Leuvensteijn, M., C. Kok-Sørensen, J. Bikker and A. van Rixtel (2008), "The impact of bank competition on the interest rate pass-through in the euro area", ECB Working Paper No.885.
- LEUVENSTEIJN, M. VAN (2008), "The Boone indicator: Identifying different regimes of competition for the American Sugar Refining Company 1890-1914", Utrecht School of Economics, Tjalling C. Koopmans Research Institute Discussion Paper Series No.08-37.
- LIN, X. AND Y. ZHANG (2009), "Bank ownership reform and bank performance in China", Journal of Banking and Finance, 33(1), 20-29.
- LIU, M. (2009), 中国银行业改革开放30年, 中国金融出版社 (in Chinese; Thirty years of reform and opening up China's banking industry, China Financial Publishing House).
- LIU, P. AND T. XIE (2006), "The monetary policy transmission in China: "Credit channel" and its limitations", FHW Berlin, Business Institute Berlin, Berlin School of Economics Working Paper No.22.
- LÖCHEL, H. AND H. LI (2011), "Understanding the high profitability of Chinese banks", Frankfurt School of Finance & Management Working Paper Series No.177.
- MA, G., Y. XIANDONG AND L. XI (2011), "China's evolving reserve requirements", BIS Working Papers No.360.
- MARTIN, M. (2012), "China's banking system: Issues for Congress", Congressional Research Service Report for Congress, 20 February.
- MATTHEWS, K., Z. XIAO AND X. ZHANG (2009), "Rational cost inefficiency in Chinese banks", Hong Kong Institute for Monetary Research Working Paper No.29/2009.
- MATTHEWS, K. AND N. ZHANG (2010), "Bank productivity in China 1997-2007: Measurement and convergence", *China Economic Review*, 21(4), 617-628.
- NORTHCOTT, C. (2004), "Competition in banking: A review of the literature", Bank of Canada, Working Paper No.2004-24. OKAZAKI, K. (2007), "Banking system reform in China: The challenges of moving toward a market-oriented economy", RAND National Security Research Division Occasional Paper No.194.
- PANZAR, J. AND J. ROSSE (1987), "Testing for 'monopoly' equilibrium", Journal of Industrial Economics, 35(4), 443–456.
- PARK, A. AND K. SEHRT (2001), "Tests of financial intermediation and banking reform in China", *Journal of Comparative Economics*, 29, 608-644.
- PEOPLE'S BANK OF CHINA (2005), "Supplement to the China Monetary Policy Report Report on Steady Progress in Market-Based Interest Rate Reform", People's Bank of China, Monetary Policy Analysis Group, January.
- PEOPLE'S BANK OF CHINA (2009), "Chinese Monetary Policy Report" (in Chinese), People's Bank of China, 2009Q2.
- PODPIERA, R. (2006), "Progress in China's banking sector reform: Has bank behaviour changed?", IMF Working Paper No.06/71.
- QIAN, J., P. STRAHAN AND Z. YANG (2011), "The impact of incentives and communication costs on information production: Evidence from bank lending", mimeograph, September.
- REZVANIAN, R. AND S. MEHDIAN (2002), "An examination of cost structure and production performance of commercial banks in Singapore", *Journal of Banking and Finance*, 26, 79-98.
- REZVANIAN, R., R. TURK ARISS AND S. MEHDIAN (2011), "Cost efficiency, technological progress and productivity growth of Chinese banking pre- and post-WTO accession, *Applied Financial Economics*, 21(7), 437-454.
- RIXTEL, A. VAN (2002), Informality and monetary policy in Japan: The political economy of bank performance, Cambridge: Cambridge University Press.
- ROSSE, J. AND J. PANZAR (1977), "Chamberlin vs Robinson: An empirical study for monopoly rents", Bell Laboratories Economics Discussion Paper No. 90.
- SALVO, A. (2010), "Inferring market power under the threat of entry: The case of the Brazilian cement industry", RAND Journal of Economics, 41(2), 326-350.
- SCHAECK, K. AND M. CIHÁK (2010), "Competition, efficiency and soundness in banking: An industrial organisation perspective", Tilburg University European Banking Center Discussion Paper No.2010-20S.
- SCHAECK, K., M. CIHÁK AND S. WOLFE (2009), "Are competitive banking systems more stable?", Journal of Money, Credit and Banking, 41(4), 711-734.

SCHMALENSEE, R. (1987), "Competitive advantage and collusive optima", International Journal of Industrial Organization, 5, 351-367.

SHAFFER, S. (1983), "The Rosse-Panzar statistic and the Lerner index in the short run", *Economics Letters*, 11(1-2), 175-178.

- (1994), "Bank competition in concentrated markets", Federal Reserve Bank of Philadelphia, Business Review, March/April, 3-16.
- SHEN, C.-H., C.-H. LU AND M.-W. WU (2009), "Impact of foreign bank entry on the performance of Chinese banks", China & World Economy, 17(3), 102-121.

SHIH, V., Q. ZHANG AND M. LIU (2007), "Comparing the performance of Chinese banks: A principal component approach", China Economic Review, 18(1), 15-34.

SINGH PRITAM SINGH, G. AND S. MUNISAMY (2008), "A cross country comparison of banking efficiency: Asia Pacific banks", *International Review of Business Research Papers*, 4(3), 73-95.

- SOEDARMONO, W., F. MACHROUH AND A. TARAZI (2013), "Bank competition, crisis and risk taking: Evidence from emerging markets in Asia", *Journal of International Financial Markets, Institutions and Money*, 23, 196-221.
- STOCK, J., J. WRIGHT AND M. YOGO (2002), "A survey of weak instruments and weak identification in Generalized Method of Moments", *Journal of Business and Economic Statistics*, 20, 518-29.
- STOCK, J. AND M. YOGO (2005), "Testing for weak instruments in linear IV regression", in: J. Stock and D. Andrews (eds.), Identification and inference for econometric models: Essays in honour of Thomas J. Rothenberg, Cambridge: Cambridge University Press, 80-108.
- SUN, J. AND N. YAMORI (2011), "The impact of local economic growth on bank profits efficiency: Evidence from city commercial banks in China", mimeograph.
- TABAK, B., D. FAZIO AND D. CAJUEIRO (2012), "The relationship between banking market competition and risk-taking: Do size and capitalization matter?", *Journal of Banking and Finance*, 36(12), 3366-3381.

TIROLE, J. (1988), The theory of industrial organisation, Cambridge, MA: MIT Press.

- UCHIDA, H.,Y. TSUTSUI (2005), "Has competition in the Japanese banking sector improved?", *Journal of Banking and Finance*, 29,419-439.
- VANDER VENNET, R. (2002), "Cost and profit efficiency of financial conglomerates and universal banks in Europe", Journal of Money, Credit and Banking, 34 (1), 254-282.
- WOLFRAM, C. (1999), "Measuring duopoly power in the British electricity spot market", American Economic Review, 89(4), 805-826.
- WONG, Y. AND M. WONG (2001), "Competition in China's domestic banking industry", Cato Journal, 21(1), 19-41.
- WORLD BANK (2012), China 2030: Building a modern, harmonious, and creative high-income society, Washington DC: The World Bank.
- XU, L. AND C.-T. LIN (2007), "Can Chinese banks compete after accession to WTO?", *Journal of Asian Economics*, 18, 883-903.
- XU, Y. (2011), "Towards a more accurate measure of foreign bank entry and its impact on domestic banking performance: The case of China", *Journal of Banking and Finance*, 35(4), 886-901.
- YAO, S., C. JIANG, G. FENG AND D. WILLENBOCKEL (2007), "On the efficiency of Chinese banks and WTO challenges", *Applied Economics*, 39, 629-643.
- YAO, S., Z. HAN AND G. FENG (2008), "Ownership reform, foreign competition and efficiency of Chinese commercial banks: A non-parametric approach", World Economy, 31(10), 1310-1326.

YEUNG, G. (2009), "How banks in China make lending decisions", Journal of Contemporary China, 18(59), 285-302.

- YI, G., (2009), "中国改革开放三十年的利率市场化进程" (in Chinese; Thirty years of reform and opening up, China's interest rate marketisation process)", Journal of Financial Research (金融研究), 1(343).
- YUAN, Y. (2006), "The state of competition of the Chinese banking industry", *Journal of Asian Economics*, 17(3), 519-534.
- ZHAO, T., B. CASU AND A. FERRARI (2010), "The impact of regulatory reforms on cost structure, ownership and competition in Indian banking", *Journal of Banking and Finance*, 34(1), 246-254.

Appendix: Measuring bank competition in China: a comparison of new versus conventional approaches applied to loan markets

Appendix A. Panzar-Rosse H-statistic

A.1. MODEL

The so-called H-statistic developed by Panzar and Rosse has been employed in a small number of empirical studies on bank competition in China (Yuan, 2006; Fu, 2009).³⁰ The Hstatistic is defined as the sum of the elasticities of a bank's total revenue with respect to that bank's input prices (Rosse and Panzar, 1977; Panzar and Rosse, 1987). Under monopoly, the revenues of the banks in question are independent of the decisions made by their actual or potential rivals. Panzar and Rosse proved that in this situation an increase in input prices will increase marginal costs, reduce equilibrium output and subsequently reduce revenues. Therefore, in this situation the H-statistic should be smaller than or equal to zero. In contrast, in the models of monopolistic competition and perfect competition, the revenue function of individual banks depends upon the decisions made by its actual or potential rivals (Bikker and Haaf, 2002). Under monopolistic competition, the change in input price is greater than the change in revenue and the H-statistic should lie between 0 and 1. Finally, under perfect competition, the H-statistic is equal to one because increases in input prices are passed on to output prices (in our case the lending rate). Higher input prices raise both marginal and average costs without, under certain assumptions, changing the optimal output of any individual bank. As some banks exit the market, the demand facing the remaining banks will increase, resulting in higher output prices and revenues equivalent to the rise in costs. Overall, a larger H-statistic indicates a higher degree of competition.

Following Bikker and Haaf (2002), we estimate the H-statistic based on the following revenue equation:

$$\ln(II_{it} / TA_{it}) = \alpha + \beta \ln(AFR_{it}) + \gamma \ln(PPE_{it}) + \delta \ln(PCE_{it}) + \eta_1 \ln(LNS_TA_{it}) + \eta_2 \ln(ONEA_TA_{it}) + \eta_3 \ln(DPS_F_{it}) + \eta_4 \ln(EQ_TA_{it}) + \eta_5 OI_II_{it} + \Sigma_{h=1...H-1} \mathcal{G}_h d_i^h + error_{it}$$
(A.1)

The dependent variable $\ln(ll_t / TA_t)$ is the logarithm of the ratio of interest income to total assets.³¹ Hence, we employ the so-called *scaled* version of the Panzar-Rosse model, in order to be able to compare our results with those of Yuan (2006) and Fu (2009). We use the ratio of interest expenses to total funding as a proxy for the average funding rate (*AFR*). The ratio of personnel expenses to total assets is adopted as a proxy for the wage rate or price of personnel expenditure (*PPE*). Furthermore, the ratio of non-interest expenses to fixed assets is used as a proxy for the price of capital expenditure (*PCE*). The H-statistic, or the sum of the elasticities of a bank's total revenue with respect to that bank's input prices, is then defined as $H=\beta+\gamma+\delta$.

³⁰. Bikker et al. (2007) and Bikker and Spierdijk (2008) include China in Panzar-Rosse based investigations of bank competition in large country samples as well.

³¹. Bikker et al. (2007) and Bikker et al. (2012) demonstrate that taking interest income as share of total assets, or the inclusion of scale variables as explanatory variables, may lead to overestimate competition and distorted tests results. Instead, they suggest using unscaled variables, ie using interest income, as the dependent variable. However, we use the scaled version of the H-statistic in order to be able to compare our results with those of Yuan (2006) and Fu (2009). As a robustness check, we also have estimated unscaled H-statistic. For more details see Appendix E (E.1).

We follow the standard approach to include several bank specific variables as control variables to capture bank differences in risk, size and business structure. As the H-statistic assesses market structure by evaluating the relationship between costs and revenues, bank-specific characteristics need to be controlled for. We take the following variables into account: The ratio of loans to total assets (*LNS_TA*); the ratio of other non-earning assets to total assets (*ONEA_TA*) reflects the composition of assets; the ratio of customer deposits to the sum of customer deposits and short-term funding (*DPS_F*) captures the features of the funding mix; the ratio of equity to total assets (*EQ_TA*) is employed to reflect risk; the ratio of other income to interest income (*OI_II*) proxies the specific business structure. The variable $d/^{p}$ is the bank type dummy. As we have four types of banks in our sample (SOCB, JSCB, CCB and FOREIGN), we drop the *CCB* dummy to avoid over identification. The respective data are summarised in Table 3.

The coefficient for *LNS_TA* is expected to be positive, as more lending potentially generates more interest income. The coefficient for *ONEA_TA* may be negative, as a higher ratio may be associated with lower interest income. *OI_II* is likely to have a negative coefficient, because generating other income might come at the expenses of interest income. For the signs of the coefficients for the other control variables, no prior expectations are offered by theory.

An important limitation of the H-statistic is that the market must be in long-run equilibrium, ie the return on total assets (ROA) should not be significantly correlated with input prices. The underlying motivation is that competitive markets will equalise the risk-adjusted rates of return across firms to such an extent that, in equilibrium, their correlation with input prices will be zero (Gutiérrez de Rozas, 2007). As is standard in the literature, we test the long-run equilibrium condition based on a regression in which the dependent variable is In(ROA), while the independent variables are the same as in Equation (A.1):

$$\ln(ROA_{it}) = \alpha + \beta \ln(AFR_{it}) + \gamma \ln(PPE_{it}) + \delta \ln(PCE_{it}) + \eta_1 \ln(LNS_TA_{it}) + \eta_2 \ln(ONEA_TA_{it}) + \eta_3 \ln(DPS_F_{it}) + \eta_4 \ln(EQ_TA_{it}) + \eta_5 OI_II_{it} + \Sigma_{h=1...h-1}G_h d_i^h + error_{it}$$
(A.2)

where ROA is defined as net income over total assets. With this specification, $E=\beta+\gamma+\delta=0$ indicates long-run equilibrium, while E<0 represents disequilibrium.

A.2. EMPIRICAL RESULTS PANZAR-ROSSE H-STATISTIC

Estimations are carried out with recursive least squares.³² This approach does not impose any parametric structure on the evolution of the H-statistic and has the advantage of allowing for the assessment of bank competition for various time windows in our sample. We do not employ the commonly used yearly estimation of the H-statistic, as in Fu (2009) and Yuan (2006), because the test statistics based on a small number of banks in the early years of our sample might not be reliable. Another advantage of recursive least squares is that this approach can avoid the erratic pattern of the H-statistic which is often obtained with yearly estimations (Bikker and Spierdijk, 2008). We estimate Equation (A.1) recursively, starting with a window of two years and expanding the sample by one year at a time. In total we obtain 12 estimation windows. The results are summarised in Table A.1, Panel A. To ensure standard errors and statistics that are robust to arbitrary heteroskedasticity and autocorrelation, kernel-

^{32.} Bikker and Spierdijk (2008) employ a parametric approach by incorporating time variant coefficients in the revenue equation. We use this approach as one of the robustness tests in Appendix E (E.2). We also tested 3-year rolling-window regressions and found similar results to recursive least squares. Results are available upon request.

based heteroskedastic and autocorrelation consistent (HAC) variance estimations are employed. The long-run equilibrium condition tests are provided for each time window, which are summarised in Panel B of Table A.1. To save space, the coefficients of the control variables are not reported. Nevertheless, the signs of the coefficients of the control variables confirm our prior expectations.³³

The estimated H-statistic show a slightly increasing level of bank competition for the early time windows, but with an increasing time span, bank competition generally follows a declining pattern. This result is rather similar to those obtained by Yuan (2006) and Fu (2009). However, it should be noted that the differences between the H-statistic across all time windows are not statistically different. Wald F-tests on the sum of the input price elasticities reject both H=1 (perfect competition) and H=0 (monopoly), indicating that all time windows can be characterised by monopolistic competition. Long-run equilibrium condition tests are rejected for all time windows except for one.

To assess whether bank competition experienced structural changes, we estimate H-statistic for the whole sample and two subsamples. The break year for the subsamples is 2001, the year of WTO accession, resulting in the pre-WTO period 1996-2001 and post-WTO period 2002-2008.³⁴ The results for the H-statistic are reported in Table A.2, while the long-run market equilibrium condition tests for the whole sample and sub-samples are reported in Table A.3. The H-statistics for each sub-sample and for the whole sample again suggest that Chinese banking markets were in a state of monopolistic competition. When comparing the H-statistic for each subsample, we cannot reject the null hypothesis that they are equal across the subsamples for any conventional significance level, suggesting no significant structural change. Table A.3 shows that the long-run equilibrium condition (E=0) is rejected for the whole sample period and both subsample periods. This is likely to be related to the ongoing process of financial reform in China, which makes it unlikely that the banks have fully adjusted to market conditions. Hence, inferring competitive conditions from these results for China are likely to be biased.³⁵

To conclude, using similar specifications as Yuan (2006) and Fu (2009), we find that the market structure indicated by our results is that of monopolistic competition. Moreover, the level of competition does not change significantly across time. Finally, it should be noted that the long-run equilibrium condition underlying the Panzar-Rosse model generally is not satisfied.³⁶

A.3. BIAS PANZAR-ROSSE H-STATISTIC DUE TO INTEREST RATE REGULATION

Feyzioğlu et al. (2009) and Bikker et al. (2007) indicate that the H-statistic probably picks up the co-movement of regulated deposit and lending rates in China. So, instead of measuring the degree of pass-through of input prices to output prices that would measure the degree of competition in a liberalised market, it measures the degree in which the regulator sets deposit and lending rates jointly. The H-statistic may be biased upwards due to the high correlation

^{33.} Table A.2 reports the coefficients of the various control variables. The positive sign for *LNS_TA* and the negative signs for *ONEA_TA* and *OI_II* confirm our prior expectations.

^{34.} The selection of 2001 as break year in the dataset is supported by formal structural break tests.

^{35.} To test for monopolistic or perfect competition, it is necessary for the observations to be generated in long-run equilibrium (Panzar-Rosse, 1987). This equilibrium may not have been achieved yet in transitional economies, doubting its usefulness to assess competition in these markets (Shaffer, 1994; Northcott, 2004).

^{36.} We demonstrate in Appendix E (E.1) by using the unscaled version of the H-statistic, which is theoretically more sound than the scaled version (see Bikker et al., 2007, and Bikker et al., 2012), that the market structure for the pre-WTO period featured perfect competition and that for the post-WTO period monopolistic competition.

between the ceilings on deposit and loan rates, which may have been especially relevant for the earlier sample years when interest rate deregulation had hardly started. The high values of the H-statistic for the pre-WTO period reported in previous studies (Yuan, 2006; Fu, 2009) and in our own estimates in the previous section likely are driven by this effect. The ceiling on the lending rate was abolished in 2004, which may have reduced the impact of this bias in subsequent years. This conclusion is supported by the findings reported in Table A.2, where the coefficient of the average funding rate (AFR) is much higher in the pre-WTO period, while dropping considerably later on when the lending rate ceiling was abolished.

H-statistic and long-run equilibrium condition: Recursive least squares									
Panel A: H-Stati	stic								
	In(AFR)	ln(PPE)	In(PCE)	н	H ₀ : H=1 chi ² (1)	H ₀ : H=0 chi ² (1)	Nr.obs	F	Adj R ²
1996–1997	0.717***	0.0736**	-0.0647*	0.7254	7.77***	54.24***	25	113.09	0.836
1996–1998	0.778***	0.0652**	-0.0588*	0.7840	7.48***	98.52***	39	66.13	0.864
1996–1999	0.715***	0.0771**	-0.0493	0.7424	20.64***	171.51** *	60	64.37	0.821
1996–2000	0.689***	0.0828**	-0.026	0.7461	26.48***	228.61**	84	92.34	0.852
1996–2001	0.650***	0.0986** *	-0.024	0.7246	38.79***	268.5/**	112	74.17	0.852
1996–2002	0.550***	0.113***	0.00124	0.6642	57.99***	220.02 ^{~~} * 245 21**	144	44.51	0.858
1996–2003	0.535***	0.136***	0.0113	0.6818	53.43***	*	184	51.39	0.837
1996–2004	0.517***	0.129***	0.0303	0.6757	52.01***	225.9***	223	51.58	0.826
1996–2005	0.512***	0.134***	0.0164	0.6627	60.64***	234.14** * 250 59**	277	62.03	0.823
1996–2006	0.507***	0.120***	0.0097	0.6364	81.79***	*	350	74.11	0.799
1996–2007	0.522***	0.131***	0.0121	0.6643	74.54***	291.9***	432	86.4	0.795
						360.86**			
1996–2008	0.532***	0.126***	0.0183	0.6765	82.5***	*	493	96.5	0.777
Panel B: Long-r	un equilibriur	n condition	test						
	In(AFR)	In(PPE)	In(PCE)	Н	H ₀ : E=1 chi ² (1)	Equilibri um	Nr.obs	F	Adj R ²
1996–1997	-0.0189	-0.0589	-0.0948	-0.1726	0.31	А	24	13.17	0.528
1996–1998	1.186***	-0.164	-0.163	0.8590	6.14**	R	38	16.72	0.585
1996–1999	0.852***	-0.121	-0.128	0.6026	9.71***	R	59	9.904	0.364
1996–2000	0.795***	-0.0983	-0.0735	0.6232	15.83***	R	83	11.82	0.389
1996–2001	0.566***	-0.0499	0.0414	0.5573	15.83***	R	111	8.406	0.345
1996–2002	0.341***	0.00301	0.112	0.4556	10.92***	R	141	7.702	0.307
1996–2003	0.362***	-0.0416	0.0621	0.3825	7.25***	R	181	6.391	0.263
1996–2004	0.311***	-0.0413	0.0174	0.2871	4.00**	R	219	4.969	0.203
1996–2005	0.283***	-0.0695	0.0494	0.2625	3.9**	R	273	5.17	0.167
1996–2006	0.235***	-0.0917	0.0616	0.2049	2.74*	R	345	5.86	0.145
1996-2007	0.267***	-0.0789	0.131**	0.3193	7.34***	R	427	7.847	0.167
1996–2008	0.286***	-0.05	0.155**	0.3907	14.9***	R	486	9.661	0.182

* represents significance level of 10%, ** represent significance level of 5%, *** represent significance level of 1%. A and R represent "Accepting" and "Rejecting" the null hypothesis that E=0 (equilibrium) at a 10% significance level.

H-statistic point estimates: Whole sample and subsamples Table A.2												
	199	6–2008	199	96–2001	200	02–2008						
ln(AFR)	0.532***	(24.63)	0.650***	(20.09)	0.537***	(21.81)						
ln(PPE)	0.126***	(4.89)	0.0986***	(3.85)	0.145***	(4.03)						
ln(PCE)	0.0183	(1.27)	-0.024	(-0.96)	0.0149	(0.92)						
lnLNS_TA	0.0920*	(1.68)	0.0293	(0.60)	0.0905	(1.41)						
LnONEA_TA	-0.0191***	(-3.75)	-0.0545***	(-4.31)	-0.0140***	(-2.62)						
InDPS_F	0.117***	(2.61)	-0.0378	(-1.03)	0.179***	(3.88)						
lnEQ_TA	0.0846***	(3.62)	0.120***	(3.55)	0.0841***	(3.08)						
lnOI_II	-0.0737***	(-9.56)	-0.0736***	(-5.56)	-0.0760***	(-8.48)						
SOCB	-0.0779***	(-2.83)	-0.100**	(-1.98)	-0.0485	(-1.59)						
JSCB	-0.0137	(-0.58)	0.0870**	(2.10)	-0.0595*	(-1.94)						
FOREIGN	-0.204***	(-4.29)	-0.402***	(-3.59)	-0.163***	(-3.34)						
Constant	-0.361**	(-2.23)	-0.252	(-1.30)	-0.207	(-0.99)						
H-statistic	0.67	65	0.72	46	0.69	74						
H ₀ : H=0 chť ² (1)	360.3	86***	268.	57***	226.	37***						
H ₀ : H=1 chi ² (1)	82.	50***	38.	79***	42.	63***						
H _{prewto} =H _{postwto}	chi ² (1)=0.22	p-value=0.6357										
Nr. Obs	49	93	1	12	381							
F	96.	50***	74.	17***	83.00***							
Adi R ²	0.7	77	0.8	52	0.768							

z-values in parenthesis; * represents significance level of 10%, ** represents significance level of 5%,

*** represents significance level of 1%

Dependent variable: InROA Table A.3								
	199	96–2008	199	96–2001	200	2002–2008		
ln(AFR)	0.286***	(3.42)	0.566***	(3.25)	0.341***	(3.70)		
ln(PPE)	-0.05	(-0.61)	-0.0499	(-0.36)	-0.0621	(-0.62)		
ln(PCE)	0.155**	(2.57)	0.0414	(0.38)	0.119**	(1.98)		
lnLNS_TA	-0.137	(-0.76)	-0.875*	(-1.67)	-0.00647	(-0.03)		
LnONEA_TA	-0.111***	(-5.35)	-0.165***	(-2.83)	-0.107***	(-4.98)		
lnDPS_F	0.142	(1.01)	-0.113	(-0.77)	0.403***	(3.07)		
lnEQ_TA	0.355***	(4.25)	0.473***	(2.65)	0.355***	(3.77)		
lnOI_II	-0.00345	(-0.13)	-0.0633	(-1.47)	-0.00194	(-0.07)		
SOCB	-0.211*	(-1.76)	-0.614**	(-2.41)	0.0405	(0.36)		
JSCB	-0.0959	(-1.08)	0.0739	(0.43)	-0.234**	(-1.98)		
FOREIGN	-0.486***	(-2.64)	-1.676***	(-2.82)	-0.21	(-1.41)		
Constant	-3.155***	(-5.62)	-2.802***	(-2.94)	-2.880***	(-4.24)		
H ₀ : E=0 chť ² (1)	14.90***		15.	15.83***		10.22***		
Nr obs	486		1	111		375		
F	9.661***		8.4	06***	8.950***			
Adj R ²	0.1	82	0.3	45	0.2	08		

Long-run equilibrium condition: Whole sample and subsamples

z-values in parenthesis; * represents significance level of 10%, ** represents significance level of 5%, *** represents significance level of 1%.

Appendix B. Underlying estimations elasticity-adjusted Lerner index

	Yearly estimates		Subsample estimates		
Panel A:Cost Equation					
	Coefficient	z-value	Coefficient	z-value	
In(securities)	-0.505***	(-2.76)	-0.285	(-1.51)	
(ln(securities)) ²	0.0300***	(3.57)	0.0314***	(3.61)	
In(other services)	0.973***	(5.23)	0.831***	(4.37)	
(In(other services)) ²	0.0426***	(4.05)	0.0288***	(2.74)	
ln(wage)-ln(other expenses)	1.270***	(4.51)	1.447***	(5.20)	
(ln(wage) –ln(other expenses))²	0.151***	(5.36)	0.150***	(5.41)	
ln(funding rate)-ln(other expenses)	0.460**	(2.26)	0.285	(1.38)	
(In(funding rate) –In(other expenses)) ²	0.197***	(4.94)	0.189***	(4.94)	
(ln(wage)ln(other expenses))*(ln(funding rate)-ln(other expenses))	-0.274***	(-4.96)	-0.268***	(-5.05)	
ln(securities) * ln(other services)	-0.0265	(-1.59)	-0.0220	(-1.32)	
ln(securities)*(ln(funding rate)-ln(other expenses))	0.0528**	(2.29)	0.0415*	(1.84)	
ln(securities)*(ln(wage)-ln(other expenses))	-0.164***	(-5.31)	-0.133***	(-4.25)	
ln(other services)*(ln(funding rate)-ln(other expenses))	-0.00508	(-0.21)	-0.0306	(-1.32)	
ln(other services) *(ln(wage)-ln(other expenses))	0.147***	(4.66)	0.161***	(5.19)	
ln(equity/assets)	-0.0116	(-0.06)	0.0321	(0.17)	
(ln(eqauity/asset)) ²	-0.00769	(-0.23)	0.000250	(0.01)	
SOCB	0.398***	(3.11)	0.371***	(3.04)	
JSCB	0.332***	(4.37)	0.304***	(4.51)	
ССВ	0.194***	(3.25)	0.189***	(3.44)	
constant	4.054***	(4.17)	4.273***	(4.42)	
Panel B:Supply Equation					
ln(loans)	0.864***	(6.39)	0.724***	(4.75)	
(ln(loans)) ²	0.0263**	(2.52)	0.0298**	(2.54)	
ln(loans) * ln(securities)	-0.0370**	(-2.35)	-0.0522***	(-3.06)	
ln(loans) * ln(other services)	-0.0432***	(-3.26)	-0.0226	(-1.55)	
ln(loans)*(ln(funding rate)-ln(other expenses))	-0.0366*	(-1.69)	0.00182	(0.08)	
ln(loans)*(ln(wage)-ln(other expenses))	0.0374	(1.54)	-0.0135	(-0.52)	
λ ₁₉₉₇	0.0616***	(9.17)			
λ_{1998}	0.0449***	(7.89)			
λ_{1999}	0.0294***	(5.89)			
λ_{2000}	0.0208***	(4.84)			
λ2001	0.0191***	(4.66)			
λ_{2002}	0.0151***	(4.19)			
λ2003	0.0167***	(4.84)			
λ_{2004}	0.0196***	(5.96)			
λ2005	0.0227***	(7.39)			
λ_{2006}	0.0255***	(9.18)			
λ2007	0.0359***	(13.99)			
λ2008	0.0415***	(13.18)			
λ1996-2001			0.0287***	(9.38)	
λ2002-2008			0.0269***	(14.12)	
Number of observations:	453		453		
z-values in parenthesis: * $n < 1$ ** $n < 0.05$ *** $n < 0.01$. Time dummies in	cost equation not	t shown to save	snace		

Chart B.1: Marginal costs and traditional and elasticity-adjusted Lerner indices



MCe and MCI are the average marginal costs used in the calculation of the elasticity-adjusted Lerner and traditional Lerner indices, respectively. The dotted line in the right panel represents 95% confidence interval of traditional Lerner index.

Appendix C. Estimation translog cost functions (TCF) for PE indicator

In order to be able to calculate marginal costs, we estimate, for each bank group, a translog cost function (TCF) using individual bank observations. This is done by allowing for bank type dummies $d/^{h}$ to interact with the independent variables in the TCF, resulting in the following form:

$$\ln c_{it}^{h} = \alpha_{0} + \Sigma_{t=1,\dots,(T-1)} \gamma_{t} d_{t} + \Sigma_{j=1,\dots,K} \delta_{jh} d_{i}^{h} \ln x_{ijt} + \Sigma_{j=1,\dots,K} \Sigma_{k=1,\dots,K} \epsilon_{jkh} d_{i}^{h} \ln x_{ijt} \ln x_{ikt} + \nu_{it}$$
(C.1)

where the dependent variable $c^{h_{it}}$ reflects the production costs of bank *i* (*i*= 1,..., *N*) in year *t* (*t* = 1,..., *T*). The sub-index *h* (*h* = 1,..., *H*) refers to the type category of the bank (state owned banks, joint-stock banks, city commercial banks, foreign banks). The variable d^{h_i} is a bank type dummy variable, which is 1 if bank *i* is of type *h* and otherwise zero. Another dummy variable is d_t , which is 1 in year *t* and otherwise zero. The coefficients α_h , δ_{jh} and e_{jkh} , all vary with *h*, the bank type. The parameters γ_t are the coefficients of the time dummies and v_{it} is the error term. The explanatory variables x_{ikt} follow the same interpretation as in Section 4.1.1. The two standard properties of TCF, linear homogeneity in input prices and cost-exhaustion, hold for each bank type *h*. Namely, Equation (C.2) holds for each bank type *h*:

$$\delta_1 + \delta_2 + \delta_3 = 1, \epsilon_{1,j} + \epsilon_{2,j} + \epsilon_{3,j} = 0 \text{ for } j = 1, 2, 3, \text{ and } \epsilon_{k,1} + \epsilon_{k,2} + \epsilon_{k,3} = 0 \text{ for } k = 4, ..., K$$
(C.2)

The marginal costs of output category j = l (of loans) for bank *i* of category *h* in year t, mc^{h}_{ilt} are defined as:

$$mc_{ilt}^{h} = \partial c_{it}^{h} / \partial x_{ilt} = \left(c_{it}^{h} / x_{ilt} \right) \partial \ln c_{it}^{h} / \partial \ln x_{ilt}$$
(C.3)

The term $\partial \ln c^{h}_{it}/\partial \ln x_{ilt}$ is the first derivative of Equation (C.1) of costs to loans. We use the marginal costs of the output component 'loans' only (and not for the other K_{1} components) as we investigate the loan markets. We estimate a separate translog cost function for each bank category (SOCB, JSCB, CCB and FOREIGN), allowing for differences in the production structure across bank types. This leads to the following equation of the marginal costs for output category loans (*I*) for bank *i* in category *h* during year *t*:

$$mc_{ilt}^{h} = c_{it}^{h} / x_{ilt} \left(\delta_{1h} + 2\epsilon_{1lh} \ln s_{ilt} + \Sigma_{k=1,\dots,K;k\neq l} \epsilon_{1kh} \ln x_{ikt} \right) d_{i}^{h}$$
(C.4)

Estimate translog cos	t function by bank type
-----------------------	-------------------------

Table C.1

SOCB			JSCB		ССВ		FOREIGN	
Dependent variable: In(costs)-In(othe	r expenses)							
Outputs								
ln(loans)	0.768**	(2.23)	1.332***	(5.15)	1.174***	(8.91)	1.759***	(17.22)
(ln(loans)) ²	-0.0743**	(-2.01)	-0.00285	(-0.07)	0.0595***	(4.11)	-0.0263**	(-2.41)
ln(securities)	0.265	(0.70)	-0.162	(-0.61)	-0.130	(-0.98)	0.0839	(0.96)
(ln(securities)) ²	0.0950***	(4.73)	0.0143	(0.53)	0.0486***	(5.24)	-0.0201***	(-3.81)
ln(other services)	0.945***	(4.76)	-0.411***	(-3.38)	0.142*	(1.82)	-0.0896	(-0.91)
(ln(other services)) ²	0.0144***	(4.21)	-0.00469	(-0.90)	0.00641*	(1.70)	-0.0371***	(-2.90)
Input prices								
ln(wage)-ln(other expenses)	2.907***	(4.78)	-0.698***	(-5.37)	0.352**	(2.04)	1.896***	(13.39)
ln(funding rate)-ln(other expenses)	0.739**	(2.15)	0.966***	(3.76)	-0.0135	(-0.08)	-1.179***	(-9.83)
(ln(wage) -ln(other expenses)) ²	-0.364***	(-8.82)	-0.00712	(-0.60)	0.0872***	(4.08)	0.111***	(5.81)
(ln(funding rate) –ln(other								
expenses)) ²	-0.0439***	(-3.11)	0.0937***	(3.79)	0.0539***	(2.69)	0.106***	(8.19)
Cross-products between input prices								
(ln(wage) —ln(other								
expenses))*(ln(funding rate)-ln(other								
expenses))	0.0831***	(2.82)	-0.0782***	(-3.00)	-0.128***	(-3.50)	-0.225***	(-7.45)
Cross-products between outputs								
ln(loans) * ln(securities)	-0.0247	(-0.47)	-0.0163	(-0.25)	-0.0947***	(-4.52)	-0.0467***	(-4.06)
ln(loans) * ln(other services)	-0.115***	(-5.40)	0.0454*	(1.91)	-0.0269**	(-2.17)	-0.00174	(-0.12)
ln(securities) * ln(other services)	-0.00459	(-0.31)	-0.0176	(-0.97)	0.0122	(0.96)	0.0810***	(5.53)
Cross-products between outputs and	l input price	S						
ln(loans)*(ln(funding rate)-ln(other								
expenses))	-0.0784**	(–2.30)	-0.00700	(-0.15)	0.0481*	(1.93)	0.216***	(9.49)
ln(loans)*(ln(wage)-ln(other								
expenses))	-0.745***	(-10.57)	0.123***	(5.19)	0.0975***	(3.80)	-0.130***	(-4.88)
ln(securities)*(ln(funding rate)-								
ln(other expenses))	0.111***	(4.26)	0.0174	(0.46)	-0.0177	(-0.99)	0.0360**	(2.18)
ln(securities)*(ln(wage)-ln(other								
expenses))	0.472***	(12.95)	-0.0769***	(-3.06)	-0.0632***	(-3.21)	-0.0811***	(-3.98)
ln(other services)*(ln(funding rate)-								
ln(other expenses))	-0.0328**	(–2.34)	-0.0119	(-0.60)	0.0222**	(2.17)	-0.198***	(-9.34)
ln(other services) *(ln(wage)-ln(other								
expenses))	-0.126***	(-8.14)	-0.0134	(-0.86)	-0.00528	(-0.43)	0.144***	(5.56)
Control variables								
ln(equity/assets)	-2.490***	(-22.49)	0.105	(0.90)	-0.0254	(-0.13)	0.795***	(5.19)
(ln(eqauity/asset)) ²	-0.371***	(–22.37)	0.0256	(1.45)	0.00136	(0.04)	0.163***	(4.96)
Constant	-0.00271	(-0.86)	-0.0657***	(-3.16)	0.000664	(0.02)	1.03e-13	(0.00)
F	1760	657.7	8666	53.1	1837	4.9	1384	9.3
Adj-R ²	0.9	997	0.99	998	0.99	90	0.9987	

Financial reform indices Table D.1											
	1996	1997	1998	1999	2000	2001	2002	2003	2004	2005	
Financial liberalization index	0.179	0.226	0.298	0.345	0.345	0.345	0.393	0.393	0.488	0.488	
Interest rate liberalization index	0	0	0	0	0	0	1	1	2	2	

Source: Abiad et al. (2010), http://www.imf.org/external/pubs/ft/wp/2008/data/wp08266.zip. A value of 0 indicates a fully repressed financial system, while a value of 1 points at a fully liberalised one. Interest rate liberalization index, which takes values of 0, 1, 2 and 3, indicates respectively a fully repressed, partially repressed, partially liberalised and fully liberalised system.

Appendix E. Additional robustness tests

In this section, we present a number of tests to check the robustness of our results for alternative specifications and estimation methods. The robustness checks show that alternative definitions of competition indicators do not change our results significantly. Specifically, we test in Appendix E whether the main results are sensitive to: 1) unscaled version of the Panzar-Rosse H-statistic; 2) parametric approach of Panzar-Rosse; 3) alternative Lerner index (conjectural variation); 4) alternative definition of PE indicator; 5) calculation marginal costs.

E.1. UNSCALED PANZAR-ROSSE H-STATISTIC

In our estimation of the Panzar-Rosse H-statistic (Appendix A), we used the scaled approach, ie the logarithm of interest income to total assets as the dependent variable, in order to be able to compare our results with those of Yuan (2006) and Fu (2009). However, we know from the literature that this approach is biased. Bikker et al. (2007) and Bikker et al. (2012) demonstrate that taking interest income as a share of total assets as the dependent variable, instead of the absolute amount of interest income (unscaled version), overestimates the degree of competition. In addition, when using this specification, results indicating both a monopoly and a situation of perfect competition will be distorted. The inclusion of scale variables as explanatory variables in the revenue function has a similar distorting effect.

As a sensitivity test, we estimate an unscaled version of the H-statistic using In(interest income) as dependent variable.³⁷ The results show even a more pronounced different pattern before and after China joined the WTO: The H-statistic indicate that Chinese loan markets were characterised by perfect competition before WTO accession and moved to monopolistic competition afterwards. Yuan (2006) and Fu (2009) reached similar conclusions, although with the scaled approach. Hence, the results of the theoretically better founded unscaled version of the Panzar-Rosse model show that Chinese loan markets were already in a state of perfect competition before further important financial reforms were implemented in the context of WTO accession in 2001 and that since then competition only declined. We hold the view that applying the more preferable unscaled version actually reinforces the shortcomings of the H-statistic for China.

E.2. PARAMETRIC APPROACH OF PANZAR-ROSSE

Bikker and Spierdijk (2008) employed a parametric approach by incorporating time variant coefficients in the revenue equation, which allows for formally testing the evolution of bank competition over time. As a robustness test, we also estimated the H-statistic assuming a parametric structure of the evolution of competition, with the following specification:

$$\ln(I_{it}/TA_{it}) = \alpha + (\beta \ln(AFR_{it}) + \gamma \ln(PPE_{it}) + \delta \ln(PCE_{it}))^{2} \exp(\zeta^{*}t) + \eta_{1} \ln(LNS_{-}TA_{it}) + \eta_{2} \ln(ONEA_{-}TA_{it}) + \eta_{3} \ln(DPS_{-}F_{it}) + \eta_{4} \ln(EQ_{-}TA_{it}) + \eta_{5}OI_{-}II_{it} + \Sigma_{h=1,H-1} \leq_{h} d_{i}^{h} + error_{it}$$
(E.2.1)

where *t* is time, and the H-statistic is defined as $H_t = (\beta + \gamma + \delta) \exp(\zeta t)$. With this specification, if $\zeta = 0$, the competitive structure is constant over time, while $\zeta > 0$ ($\zeta < 0$) indicates an increasing (decreasing) level of competitiveness over time. Estimation is carried out with nonlinear least square. Our results show a significantly negative time coefficient ζ of -0.0041

37

Results are available upon request.

(p-value 0.0000), suggesting an annual decrease in the level of competition for the whole sample period. Wald F-tests on the sum of the input price elasticities reject the H-statistic being 1 (perfect competition) and 0 (monopoly) at a 1% significance level, indicating that all years could be characterised by monopolistic competition. Furthermore, a Wald F-test on the long-run equilibrium condition rejects E=0 at a 1% significance level for each year which suggests that Chinese loan markets were in disequilibrium. These results confirm that our results for the H-statistic are not sensitive to specific estimation methods. Results are available upon request.

E.3. ALTERNATIVE LERNER INDEX (CONJECTURAL VARIATION)

In Section 6.1 we calculated the elasticity-adjusted Lerner index L by first estimating λ , ie the ratio of conjectural variation Θ to the elasticity of demand ϵ . Subsequently we could estimate L as λ/p , with p the average price of loans (average lending rate). An alternative approach is to estimate explicitly the conjectural variation Θ by simultaneously estimating the TCF (Equation 4.2), the supply equation (Equation 4.6) and an inverse loan demand function. Then the conjectural variation parameter Θ can serve as a direct measure of competition. In a perfectly competitive market, Θ_i equals to zero for all *i*, while for a monopoly it equals to one. This approach is adopted in Uchida and Tsutsui (2005) for Japanese banking market. Following this approach, we find that the estimated inverse demand elasticity is very stable across all years, which implies that conjectural variation follows a similar pattern to the evolution of the elasticity-adjusted Lerner index. Subsample estimations show that the conjectural variation is 0.068 and 0.087 for the pre-WTO respectively the post-WTO period, with the former being more competitive than the latter at a 1% significance level. We conclude that our main results obtained with the elasticity-adjusted Lerner index hold if conjectural variation is employed as a direct measure of competition. The full estimation process and results are not reported here to save space, but are available from the authors upon request.

E.4. ALTERNATIVE DEFINITION OF PE INDICATOR

We calculated the PE indicator by using the logarithm of π_{ilt} or profits obtained from loans as the dependent variable (see Section 6.2). This is a more accurate measure of profits generated by loan business. Alternatively, as a robustness check, we follow Boone et al. (2004) and use the logarithm of variable profits as the dependent variable. This approach has the advantage that it avoids potential estimation errors, as variable profits can be obtained directly from accounting data. At the same time, it has the disadvantage that variable profits capture not only profits from loans but also those from other activities. Variable profits are defined here as the difference between total income and the sum of interest expenses and other non-interest expenses.³⁸ We find that they are highly correlated with the definition of profits that we used in Section 6.2, with a Pearson correlation coefficient of 0.9607.

Similar to the other estimations, we estimate yearly and subsample PE indicators which are reported in Panels B of Table E.1 respectively E.2. Again, competition follows the same pattern that we reported for the initial results. The structural break test for the point estimates for the two subsamples again supports our finding that the pre-WTO period is less competitive than the post-WTO period.

^{38.} An alternative definition of variable profits is interest income - (interest expenses + other non-interest expenses). Our main conclusions are not sensitive to this alternative definition. Results are available upon request.

E.5. CALCULATION OF MARGINAL COSTS

For the (elasticity-adjusted) Lerner index, we assumed that the Translog Cost Function (TCF) for each bank group (SOCB, JSCB, CCB, FOREIGN) is the same, as only the constant term is allowed to vary across bank groups through bank type dummies (Equation 4.2). For the PE indicator, we improved the estimation by imposing different cost functions on different bank groups and estimated a separate TCF for each bank group. Both ways of treating cost functions for specific bank groups are generally accepted in the literature. Nevertheless, this difference could potentially generate different marginal costs. As for both the Lerner index and the PE indicator marginal cost estimations directly affect their values, it is important to test whether the contradictory results that we find could be driven by differences in the estimated marginal costs.

To this end, we conduct the following two robustness tests. First, we re-estimate the (elasticity-adjusted) Lerner indices assuming different cost functions for each bank group. Second, we re-estimate the PE indicator using the marginal costs that we estimated for the elasticity-adjusted Lerner index (MC_e), ie assuming similar translog cost functions for bank groups.

When re-estimating the (elasticity-adjusted) Lerner indices, we use different TCFs for each bank group by allowing for bank type dummies to interact with the independent variables in the TCF. We calculate again yearly and subsample values, which are shown in Table E.3. The modification in the TCF turns out to change the elasticity-adjusted Lerner index only very marginally for both the yearly and subsample estimations³⁹. Moreover, the traditional Lerner index also resembles closely our previous results. This confirms that our previous findings are robust to different calculations of marginal costs.

The results for the re-estimation of the PE indicator using the marginal costs that we estimated in order to obtain the elasticity-adjusted Lerner index (MC_e) are shown in Panel A of Table E.1 for the yearly results and of Table E.2 for the subsample results. The former follows a very similar pattern to our previous results. Moreover, also our conclusion that the pre-WTO period had much lower competition than the post-WTO era remains intact.

^{39.} Underlying estimations of elasticity-adjusted Lerner index not shown to save space. Results are available upon request.

Yearly estimates of alternative PE indicators

Table E.1

	Panel A	ent variable li	n(MCe)	Panel B: Dependent variable In(variable profits)				
		•	AP	. ,			AP	
	PE		chi ² (1)	AP	PE		chi ² (1)	AP
	Indicator	z-value	p-value	F(1,440)	Indicator	z-value	p-value	F(1,442)
1997	-2.314	(-1.53)	0.0000	18.08	6.656	(0.49)	0.4866	0.46
1998	-1.769	(-1.00)	0.0101	6.27	-2.183	(-1.26)	0.1021	2.53
1999	3.609	(1.3)	0.0000	33.08	-0.627	(-0.25)	0.0000	26.78
2000	-1.379	(-0.44)	0.0127	5.88	-0.667	(-0.37)	0.0000	17.13
2001	-5.748***	(-4.26)	0.0000	29.48	-3.086***	(-4.11)	0.0000	31.54
2002	-6.826**	(-2.20)	0.0009	10.46	-3.594***	(-2.64)	0.0000	20.2
2003	-3.754**	(-2.49)	0.0000	65.05	-4.391**	(-2.57)	0.0027	8.57
2004	-3.810**	(-2.25)	0.0000	72.28	-2.937***	(-3.13)	0.0000	58.35
2005	-1.605	(-1.41)	0.0000	95.18	-1.1	(-1.58)	0.0000	67.33
2006	-4.633***	(-2.87)	0.0001	14.46	-3.090***	(-3.28)	0.0000	20.59
2007	-3.669***	(-4.27)	0.0000	74.47	-3.264***	(-5.47)	0.0000	89.25
2008	-3.584***	(-3.93)	0.001	10.27	-1.959***	(-3.26)	0.0000	28.18
Constant	-2.511	(-1.00)			1.983	(1.13)		
Nr. Obs		46	54			46	56	
F	4.649					4.6	85	
Centered R ²		0.1	.32			0.0	961	

z-values in parenthesis; ** represent significance level of 5%, *** represent significance level of 1%; AP chi2 is the Angrist-Pischke (AP) first-stage chi-squared test: AP F is the Angrist-Pischke (AP) F-statistics. Test statistic can be compared to Stock-Yogo (2002, 2005) critical values for Cragg-Donald F statistic with K1=1. The Stock-Yogo weak ID test critical values at 10% maximal LIML size are 16.38. Year dummies are not reported here to save space.

Whole sample and subsample estimates of alternative PE indicators

Table E.2

				Panel B:Dependent variable:				
	Panel A: In	dependent varia	ble $ln(MC_e)$	$\ln(variable \ profits)$				
	1996–2008	1996–2001	2002–2008	1996–2008	1996–2001	2002–2008		
PE Indicator	-1.928***	-1.522	-3.717***	-2.023***	-1.487	-2.870***		
	(-3.81)	(-1.01)	(-5.65)	(-5.66)	(-1.64)	(-7.07)		
Time Trend	-0.0142	-0.508*	0.367***	0.0087	-0.461**	0.296***		
	(-0.34)	(-1.67)	(4.9)	(0.24)	(–2.20)	(4.45)		
Constant	1.069	4.889	-8.492***	0.516	4.540**	-5.236***		
	(0.73)	(1.63)	(-3.67)	(0.46)	(2.34)	(-3.37)		
H ₀ :prewto –postwto<=0 (p-value)		2.14* (0.071)			2.34* (0.063)			
Nr. Obs	464	91	373	466	91	375		
F	7.226	1.815	21.24	16.01	2.349	29.25		
Centered R ²	0.0247	0.0495	0.104	0.0691	0.101	0.141		
K-P rk Wald F	336.7	77.97	163.2	227.9	34.97	142.4		
K-P rk LM (p-value)	73.24(0.00)	15(0.000)	50.52(0.000)	64.78 (0.000)	13.59 (0.000)	45.18 (0.000)		

z-values in parenthesis; * represents significance level of 10%, ** represent significance level of 5%, *** represent significance level of 1%. K-P rk Wald F is the Kleibergen-Paap rk Wald F statistic. K-P rk LM is Kleibergen-Paap rk LM statistic. The Stock-Yogo weak ID test critical values at 10% maximal LIML size are 16.38

Lerner indices assuming different TCFs for each bank group							
	λ_t	Elasticity adjusted Lerner index	Lerner index	мс _е	MC		
1997	0.077	0.458	0.330	0.080	0.104		
1998	0.048	0.410	0.317	0.080	0.079		
1999	0.030	0.294	0.244	0.062	0.066		
2000	0.022	0.221	0.284	0.055	0.055		
2001	0.020	0.237	0.228	0.051	0.055		
2002	0.016	0.223	0.292	0.044	0.042		
2003	0.017	0.240	0.298	0.046	0.045		
2004	0.020	0.287	0.311	0.047	0.045		
2005	0.023	0.288	0.330	0.051	0.050		
2006	0.026	0.332	0.349	0.052	0.051		
2007	0.036	0.457	0.417	0.047	0.047		
2008	0.040	0.426	0.410	0.059	0.056		
1996–2001	0.027	0.235	0.284	0.071	0.069		
2002–2008	0.026	0.335	0.355	0.051	0.049		

H₀: Elasticity Adj Lerner prewto> = Elasticity Adj Lerner postwto : $chi^{2}(1)$ = 7.93 p-value = 0.0024

 λ_t are statistically different from zero for all year at a 1% significance level; MC_e and MC_l are average marginal cost estimated from elasticity-adjusted Lerner index and traditional Lerner index, respectively.

BANCO DE ESPAÑA PUBLICATIONS

WORKING PAPERS

- 1220 ENRIQUE ALBEROLA, LUIS MOLINA and PEDRO DEL RÍO: Boom-bust cycles, imbalances and discipline in Europe.
- 1221 CARLOS GONZÁLEZ-AGUADO and ENRIQUE MORAL-BENITO: Determinants of corporate default: a BMA approach.
- 1222 GALO NUÑO and CARLOS THOMAS: Bank leverage cycles.
- 1223 YUNUS AKSOY and HENRIQUE S. BASSO: Liquidity, term spreads and monetary policy.
- 1224 FRANCISCO DE CASTRO and DANIEL GARROTE: The effects of fiscal shocks on the exchange rate in the EMU and differences with the US.
- 1225 STÉPHANE BONHOMME and LAURA HOSPIDO: The cycle of earnings inequality: evidence from Spanish social security data.
- 1226 CARMEN BROTO: The effectiveness of forex interventions in four Latin American countries.
- 1227 LORENZO RICCI and DAVID VEREDAS: TailCoR.
- 1228 YVES DOMINICY, SIEGFRIED HÖRMANN, HIROAKI OGATA and DAVID VEREDAS: Marginal quantiles for stationary processes.
- 1229 MATTEO BARIGOZZI, ROXANA HALBLEIB and DAVID VEREDAS: Which model to match?
- 1230 MATTEO LUCIANI and DAVID VEREDAS: A model for vast panels of volatilities.
- 1231 AITOR ERCE: Does the IMF's official support affect sovereign bond maturities?
- 1232 JAVIER MENCÍA and ENRIQUE SENTANA: Valuation of VIX derivatives.
- 1233 ROSSANA MEROLA and JAVIER J. PÉREZ: Fiscal forecast errors: governments vs independent agencies?
- 1234 MIGUEL GARCÍA-POSADA and JUAN S. MORA-SANGUINETTI: Why do Spanish firms rarely use the bankruptcy system? The role of the mortgage institution.
- 1235 MAXIMO CAMACHO, YULIYA LOVCHA and GABRIEL PEREZ-QUIROS: Can we use seasonally adjusted indicators in dynamic factor models?
- 1236 JENS HAGENDORFF, MARÍA J. NIETO and LARRY D. WALL: The safety and soundness effects of bank M&As in the EU: Does prudential regulation have any impact?
- 1237 SOFÍA GALÁN and SERGIO PUENTE: Minimum wages: do they really hurt young people?
- 1238 CRISTIANO CANTORE, FILIPPO FERRONI and MIGUEL A. LEÓN-LEDESMA: The dynamics of hours worked and technology.
- 1239 ALFREDO MARTÍN-OLIVER, SONIA RUANO and VICENTE SALAS-FUMÁS: Why did high productivity growth of banks precede the financial crisis?
- 1240 MARIA DOLORES GADEA RIVAS and GABRIEL PEREZ-QUIROS: The failure to predict the Great Recession. The failure of academic economics? A view focusing on the role of credit.
- 1241 MATTEO CICCARELLI, EVA ORTEGA and MARIA TERESA VALDERRAMA: Heterogeneity and cross-country spillovers in macroeconomic-financial linkages.
- 1242 GIANCARLO CORSETTI, LUCA DEDOLA and FRANCESCA VIANI: Traded and nontraded goods prices, and international risk sharing: an empirical investigation.
- 1243 ENRIQUE MORAL-BENITO: Growth empirics in panel data under model uncertainty and weak exogeneity.
- 1301 JAMES COSTAIN and ANTON NAKOV: Logit price dynamics.
- 1302 MIGUEL GARCÍA-POSADA: Insolvency institutions and efficiency: the Spanish case.
- 1303 MIGUEL GARCÍA-POSADA and JUAN S. MORA-SANGUINETTI: Firm size and judicial efficacy: evidence for the new civil procedures in Spain.
- 1304 MAXIMO CAMACHO and GABRIEL PEREZ-QUIROS: Commodity prices and the business cycle in Latin America: living and dying by commodities?
- 1305 CARLOS PÉREZ MONTES: Estimation of regulatory credit risk models.
- 1306 FERNANDO LÓPEZ VICENTE: The effect of foreclosure regulation: evidence for the US mortgage market at state level.
- 1307 ENRIQUE MORAL-BENITO and LUIS SERVEN: Testing weak exogeneity in cointegrated panels.
- 1308 EMMA BERENGUER, RICARDO GIMENO and JUAN M. NAVE: Term structure estimation, liquidity-induced heteroskedasticity and the price of liquidity risk.
- 1309 PABLO HERNÁNDEZ DE COS and ENRIQUE MORAL-BENITO: Fiscal multipliers in turbulent times: the case of Spain.

- 1310 SAMUEL HURTADO: DSGE models and the Lucas critique.
- 1311 HENRIQUE S. BASSO and JAMES COSTAIN: Fiscal delegation in a monetary union with decentralized public spending.
- 1312 MAITE BLÁZQUEZ CUESTA and SANTIAGO BUDRÍA: Does income deprivation affect people's mental well-being?
- 1313 ENRIQUE ALBEROLA, ÁNGEL ESTRADA and DANIEL SANTABÁRBARA: Growth beyond imbalances. Sustainable growth rates and output gap reassessment.
- 1314 CARMEN BROTO and GABRIEL PEREZ-QUIROS: Disentangling contagion among sovereign CDS spreads during the European debt crisis.
- 1315 MIGUEL GARCÍA-POSADA and JUAN S. MORA-SANGUINETTI: Are there alternatives to bankruptcy? A study of small business distress in Spain.
- 1316 ROBERTO RAMOS and ENRIQUE MORAL-BENITO: Agglomeration matters for trade.
- 1317 LAURA HOSPIDO and GEMA ZAMARRO: Retirement patterns of couples in Europe.
- 1318 MAXIMO CAMACHO, GABRIEL PEREZ-QUIROS and PILAR PONCELA: Short-term forecasting for empirical economists. A survey of the recently proposed algorithms.
- 1319 CARLOS PÉREZ MONTES: The impact of interbank and public debt markets on the competition for bank deposits.
- 1320 OLYMPIA BOVER, JOSE MARIA CASADO, SONIA COSTA, PHILIP DU CAJU, YVONNE MCCARTHY, EVA SIERMINSKA, PANAGIOTA TZAMOURANI, ERNESTO VILLANUEVA and TIBOR ZAVADIL: The distribution of debt across euro area countries: the role of Individual characteristics, institutions and credit conditions.
- 1321 BRINDUSA ANGHEL, SARA DE LA RICA and AITOR LACUESTA: Employment polarisation in Spain over the course of the 1997-2012 cycle.
- 1322 RODOLFO G. CAMPOS and ILIANA REGGIO: Measurement error in imputation procedures.
- 1323 PABLO BURRIEL and MARÍA ISABEL GARCÍA-BELMONTE: Meeting our D€STINY. A Disaggregated €uro area Short Term INdicator model to forecast GDP (Y) growth.
- 1401 TERESA SASTRE and FRANCESCA VIANI: Countries' safety and competitiveness, and the estimation of current account misalignments.
- 1402 FERNANDO BRONER, ALBERTO MARTIN, AITOR ERCE and JAUME VENTURA: Sovereign debt markets in turbulent times: creditor discrimination and crowding-out effects.
- 1403 JAVIER J. PÉREZ and ROCÍO PRIETO: The structure of sub-national public debt: liquidity vs credit risks.
- 1404 BING XU, ADRIAN VAN RIXTEL and MICHIEL VAN LEUVENSTEIJN: Measuring bank competition in China: a comparison of new versus conventional approaches applied to loan markets.

BANCO DE ESPAÑA

Eurosistema